Master Perfection Suite[™]

Version 1.2

Software User's Guide



DCN 2.1

INSTALL LICENSE AGREEMENT

This legal document is an agreement between you, the end user ("YOU" or "LICENSEE"), and BIAS, Inc. BY CLICKING "INSTALL" ON THE FOLLOWING DIALOG, YOU ARE AGREEING TO BECOME BOUND BY THE TERMS OF THIS AGREEMENT, WHICH INCLUDES THE SOFTWARE LICENSE AND THE SOFTWARE DISCLAIMER OF WARRANTY (collectively the "Agreement"). CLICK "QUIT" ON THE FOLLOWING DIALOG IF YOU DO NOT ACCEPT THE TERMS OF THIS AGREEMENT. THIS AGREEMENT CONSTITUTES THE COMPLETE AGREEMENT BETWEEN YOU AND BIAS, INC. IF YOU DO NOT AGREE TO THE TERMS OF THIS AGREEMENT, DO NOT USE OR REGISTER THE SOFTWARE AND PROMPTLY RETURN THE SOFTWARE AND THE OTHER MATERIALS (INCLUDING WRITTEN MATERIALS, BINDERS OR OTHER CONTAINED) THAT ARE PART OF THIS PRODUCT TO THE PLACE WHERE YOU OBTAINED THEM FOR A FULL REFUND. YOUR USE OF THIS SOFTWARE SHALL CONSTITUTE ACCEPTANCE OF THIS AGREEMENT, AND ONCE YOU HAVE REGISTERE THE SOFTWARE SOFTWARE YOU MAY NOT RETURN IT.

BIAS SOFTWARE LICENSE

- GRANT OF LICENSE. In consideration of payment of the license fee, which is part of the price you paid for this product, BIAS, as Licensor, grants to you, LICENSEE, a nonexclusive license to use and display this copy of a BIAS software program (the "SOFTWARE") and accompanying user documentation ("DOCUMENTATION"). LICENSEE may install and use the SOFTWARE on only one COMPUTER at one time, and such SOFTWARE may be used by only one user at a time. BIAS reserves all rights not expressly granted to LICENSEE.
- AUTHORIZATION KEY. You must register this SOFTWARE within 14 days of installation either online (http://www.bias-inc.com/support/register/), via fax (+1 707-782-1874), or phone (+1 707-782-1866). If the SOFTWARE has not been registered within 14 days of installation, it will cease to function until registered. Once registered, you cannot return the SOFTWARE.
- 3. OWNERSHIP OF SOFTWARE AND DOCUMENTATION. As LICENSEE, you own the media on which the SOFTWARE and DOCUMENTATION are originally or subsequently recorded or fixed, but BIAS retains title and ownership of the SOFTWARE and DOCUMENTATION and all subsequent copies of the SOFTWARE and DOCUMENTATION, regardless of the form or media on which the original and other copies may exist. This License is not a sale of the original SOFTWARE or any copy or sale of the original DOCUMENTATION.
- 4. COPY RESTRICTIONS. This SOFTWARE and the DOCUMENTATION are copyrighted. Unauthorized copying of the SOFTWARE, including SOFTWARE that has been modified, merged, or included with other software, or of the DOCUMENTATION is expressly forbidden. You may be held legally responsible for any copyright infringement that is caused or encouraged by your failure to abide by the terms of this License. Subject to these restrictions, and if the SOFTWARE is not copy-protected, you may make one (1) copy of the SOFTWARE solely for backup purposes. You must reproduce and include the copyright notice on the backup copy.
- 5. USE RESTRICTIONS. As the LICENSEE, you may physically transfer the SOFTWARE from one computer to another provided that the SOFTWARE is used on only one computer at a time. You may not electronically transfer the SOFTWARE from one computer to another over a network. You may not distribute copies of the SOFTWARE or the DOCUMENTATION to others. You may not modify, adapt, translate, reverse engineer, decompile, disassemble, or create derivative works based on the SOFTWARE. You may not modify, adapt, translate, or create derivative works based on the DOCUMENTATION without the prior written consent of BIAS.

- 6. TRANSFER RESTRICTIONS. This SOFTWARE is licensed only to you, LICENSEE, and may not be assigned or otherwise transferred to anyone, whether voluntary or by operation of law, without the prior written consent of BIAS. Any authorized transferee of the SOFTWARE must agree to be bound by the terms and conditions of this Agreement, or the transfer shall be invalid. In no event may you transfer, assign, rent, lease, sell, or otherwise dispose of the SOFTWARE on a temporary or permanent basis except as expressly provided herein.
- 7. LICENSE TO U.S. GOVERNMENT. If LICENSEE is an agency or instrumentality of the United States Government, the SOFTWARE and DOCUMENTATION are "commercial computer software" and "commercial computer software documentation", and pursuant to FAR 12.212 or DFARS 227.7202, and their successors, as applicable, use, reproduction and disclosure of the Software and Documentation are governed by the terms of this Agreement.
- TERMINATION. This License is effective until terminated. This License will terminate automatically without notice from BIAS if you fail to comply with any provision of this License. Upon termination you shall destroy the DOCUMENTATION and all copies of the SOFTWARE, including modified copies, if any.
- UPDATE POLICY. BIAS may create, from time to time, updated versions of the SOFTWARE. At its option, BIAS will make such updates available to LICENSEE and transferees who have paid the update fee and returned the registration card to BIAS.
- 10. BIAS PRODUCT SUPPORT.

Unlimited Complimentary Self Help Options

If you have questions about installing or using your BIAS products, please check the manual or user guide first, you will find answers to most of your questions there. If you need further assistance, please check:

http://www.bias-inc.com/support.html-for frequently asked questions, an extensive database of related articles, troubleshooting tips, and more.

Limited Complimentary Technical Assistance

Ninety days of technical support assistance for up to three incidents is available with the initial purchase of Type 1 BIAS products (Peak Pro, Peak Pro XT, Deck, SoundSoap Pro, Master Perfection Suite). Ninety days of technical support assistance for a single incident is available with the initial purchase of Type 2 BIAS products (Peak LE, Deck LE, SoundSoap 2). Type 3 products (OEM versions) are excluded. Thirty days of technical support assistance for a single incident is available with the purchase of paid upgrades to Type 1 and Type 2 BIAS products. The support period begins when the product authorization code (PAC) is initially issued following the product registration. To receive e-mail or telephone technical support, you will need a valid serial number and must be registered for a current BIAS product. Please keep your original product serial number and PAC in a safe place as it will be required for support. For customers preferring e-mail assistance, please use the email form available to a current BIAS product.

http://www.bias-inc.com/support.html

Or, email: support@bias-inc.com

For customers preferring telephone assistance, (connect and toll charges apply), please have your computer running and close by the phone and call us at: +1-707-782-1865

(9AM-5PM Monday-Friday PST - excluding BIAS holidays)

Fee-based Technical Support Options

Fee-based Support is available for those customers whose support period has expired or for Type 3 (OEM version) customers. Please call:

+1-707-782-1865

(9AM-5PM Monday-Friday PST - excluding BIAS holidays)

For customers living outside of North America, please contact your local BIAS distributor, or contact BIAS directly. A list of BIAS distributors may be found at: http://www.bias-inc.com/distributors.html

BIAS Support Policy and Hours

BIAS, Inc. support policy and hours are subject to change without notice.

Sales and Non-Technical Customer Service

For product information, sales, and other non-technical customer service, please email or call:

sales@bias-inc.com

US: 1-800-775-BIAS (2427)

International: +1-707-782-1866

(9AM-5PM Monday-Friday PST - excluding BIAS holidays)

No technical support is available at the numbers or email address above.

11. MISCELLANEOUS. This Agreement is governed by the law of the State of California.

DISCLAIMER OF WARRANTY AND LIMITED WARRANTY

THE SOFTWARE AND DOCUMENTATION (INCLUDING INSTRUCTIONS FOR USE) ARE PROVIDED "AS IS" WITHOUT WARRANTIES OF ANY KIND. FURTHER, BIAS DOES NOT WARRANT, GUARANTEE, OR MAKE ANY REPRESENTATIONS REGARDING THE USE, OR THE RESULTS OF THE USE, OF THE SOFTWARE OR DOCUMENTATION IN TERMS OF CORRECTNESS, ACCURACY, RELIABILITY, CURRENTNESS, OR OTHERWISE. THE ENTIRE RISK AS TO THE RESULTS AND PERFORMANCE OF THE SOFTWARE IS ASSUMED BY YOU, AND NOT BIAS OR ITS DEALERS, DISTRIBUTORS, AGENTS, OR EMPLOYEES, YOU AGREE TO ASSUME THE ENTIRE COST OF ALL NECESSARY SERVICING, REPAIR, OR CORRECTION.

BIAS warrants to the original LICENSEE that the media upon which the SOFTWARE is recorded and the BIAS Key (USB dongle), if applicable, are free from defects in materials and workmanship under normal use and service for a period of ninety (90) days from the date of delivery as evidenced by a copy of your receipt. Further, BIAS hereby limits the duration of any implied warranty(ies) on the media to the respective periods stated above. Some states do not allow limitations on duration of an implied warranty, so the above limitation may not apply to you.

BIAS' entire liability and your exclusive remedy as to the media shall be, at BIAS' option, either (a) return of the purchase price or (b) replacement of the media that does not meet BIAS' Limited Warranty and which is returned to BIAS with a copy of the receipt. If failure of the media has resulted from accident, abuse, or misapplication, BIAS shall have no responsibility to replace the media or refund the purchase price. Any replacement media will be warranted for the remainder of the original warranty period or thirty (30) days, whichever is longer.

THE ABOVE ARE THE ONLY WARRANTIES OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING BUT NOT LIMITED TO THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE, THAT ARE MADE BY BIAS CONCERNING THE BIAS PRODUCT. NO ORAL OR WRITTEN INFORMATION OR ADVICE GIVEN BY BIAS, ITS DEALERS, DISTRIBUTORS, AGENTS, OR EMPLOYEES SHALL CREATE A WARRANTY OR IN ANY WAY INCREASE THE SCOPE OF THIS WARRANTY, AND YOU MAY NOT RELY ON ANY SUCH INFORMATION OR ADVICE. THIS WARRANTY GIVES YOU SPECIFIC LEGAL RIGHTS. YOU MAY HAVE OTHER RIGHTS WHICH VARY FROM STATE TO STATE.

NEITHER BIAS NOR ANYONE ELSE WHO HAS BEEN INVOLVED IN THE CREATION, PRODUCTION, OR DELIVERY OF THIS PRODUCT SHALL BE LIABLE FOR ANY DIRECT, INDIRECT, CONSEQUENTIAL, OR INCIDENTAL DAMAGES (INCLUDING DAMAGES FOR LOSS OF BUSINESS PROFITS, BUSINESS INTERRUPTION, LOSS OF BUSINESS INFORMATION, AND THE LIKE) ARISING SUCH DAMAGES. BECAUSE SOME STATES DO NOT ALLOW THE EXCLUSION OR LIMITATION OF LIABILITY FOR CONSEQUENTIAL OR INCIDENTAL DAMAGES, THE ABOVE LIMITATION MAY NOT APPLY TO YOU.

This Disclaimer of Warranty and Limited Warranty is governed by the laws of the State of California.

Should you have any questions concerning this Agreement, or if you desire to contact BIAS for any reason, please contact in writing:

BIAS Customer Sales and Service 140 Keller Street Petaluma, CA 94952 USA

Copyright 1996-2007 BIAS, Inc.

Credits:

Master Perfection Suite Software: Steve Berkley, Eric Eizenman, Guillermo Garcia, Wayne Jackson, Andrew Kimpton, David Ralley, Rick Sustek, Eric Johnson, Dan Spirlock, Ben Ventura

Audio Units ${}^{\rm TM}$, Mac OS X ${}^{\rm TM}$ and Macintosh ${}^{\rm TM}$ are registered trademarks of Apple Computer, Inc.

VST™ is a registered trademark of Steinberg Media Technologies, AG.

RTAS[™] and AudioSuite[™] are registered trademarks of Digidesign Inc./Avid Inc.

Windows XP™ and Windows Vista™ are registered trademarks of Microsoft Inc.

User's Guide: Zac Wheatcroft

Special Thanks to:

Larry the O, Ben Hester, Jason Davies, Guillermo Garcia, Christine Berkley, Paul Knowles, Christopher Denman, Gabe Khofri, Ben Ventura, Lisa Stefan, Victor Cebreros, Cory Stück, Brian Smiley, Melissa Farinha, Kristen Jacobs, Kelly Evans, Megan Peterschmidt, Larry Nilsen, Michael Van Landingham, Andrew Stone, Brent Hurtig, Ian & Christina Nyquist, Steve Cerilli, Ernest Cholakis, Bil Bryant, Howard Frederics, CKSDE, Michael Anse, Michael Barrett, Marvin Bass, Chris Borgia, Daniel Brennan, Sean Charles, Donnie Christian, Barry Diament, Jeff Dinces, Dr. Fred Bashour, Jason Fernandez, Scott Frankfurt, Chris Griffin, Barry Higgins, Gary Hobish, James L. Jackson, Julian Kwasneski, Drew Lavyne, Jeremy Lee, George Leger III, Ray Legnini, Ken Loge, Elvio Magliocca, Darin Marshall, Peter McConnell, Arnaud Melese, Jim Miller, Heather Perkins, Paul Pierron, Fernando Rodriguez, Marc Rosenberg, Brad Sarno, Jeremy Scidmore, John O. Senior, Luke Slater, William Storkson, Craig Sylvern, Rudy Trubitt, James Wadell, Jim Wintermyre, J. Mark Andrus, Michael Barrett, Robbie Berman, Bruce Bennett, Ray Brunelle, BT, Will Carpmill, Chris Clanin, Luke DeLalio, Dennie Edwards, Tom Erbe, Terje Finstad, Daniel Fisher, Kelly Fitz, Mikail Grahm, Brian Hamilton, Tom Hays, Morris Hayes, Matt Isaacson, Carson Kendall, Peer Landra, Igor Len, Randy M., Jose Ignacio Martin, Christophe Masson, Riccardo Mazza, Ron Macleod, David Mash, Peter McCabe, David McIntyre, Melvin & Milo Mole, Jeff Moore, Itai Neoran, NIN, Prayson Nowlan, Martin O'Donnell, Dan Phillips, Scott Plunkett, Gene Porfido, Les Quindipan, Ralph, Bob Rice, Brad Sarno, Arne Schulze, Seva, Eli Shauriki, Ronen Shloush, Craig Snoke, Alex Stahl, M. Stevens, Morten Stjernholm, Mark Tinley, David Torn, Rudy Trubitt, Tsvika, Earl Vickers, Charles Wiltgen, Michael Winter, and Tsuyoshi Yasukawa.

BIAS Product Support:

Unlimited Complimentary Self Help Options

If you have questions about installing or using your BIAS products, please check the manual or user guide first, you will find answers to most of your questions there. If you need further assistance, please check:

http://www.bias-inc.com/support.html

for frequently asked questions, an extensive database of related articles, troubleshooting tips, and more.

Limited Complimentary Technical Assistance

Ninety days of technical support assistance for up to three incidents is available with the initial purchase of Type 1 BIAS products (Peak Pro, Peak Pro XT, Deck, SoundSoap Pro, Master Perfection Suite). Ninety days of technical support assistance for a single incident is available with the initial purchase of Type 2 BIAS products (Peak LE, Deck LE, SoundSoap 2). Type 3 products (OEM versions) are excluded. Thirty days of technical support assistance for a single incident is available with the purchase of paid upgrades to Type 1 and Type 2 BIAS products. The support period begins when the product authorization code (PAC) is initially issued following the product registration. To receive email or telephone technical support, you will need a valid serial number and must be registered for a current BIAS product. Please keep your original product serial number and PAC in a safe place as it will be required for support. For customers preferring e-mail assistance, please use the email form available on our web site at:

http://www.bias-inc.com/support.html

Or, email:

support@bias-inc.com

For customers preferring telephone assistance, (connect and toll charges apply), please have your computer running and close by the phone and call us at:

+1-707-782-1865

9AM-5PM Monday-Friday PST - excluding BIAS holidays

Fee-based Technical Support Options

Fee-based Support is available for those customers whose support period has expired or for Type 3 (OEM version) customers. Please call:

+1-707-782-1865

9AM-5PM Monday-Friday PST - excluding BIAS holidays

For customers living outside of North America, please contact your local BIAS distributor, or contact BIAS directly. A list of BIAS distributors may be found at:

http://www.bias-inc.com/distributors.html

BIAS Support Policy and Hours

BIAS, Inc. support policy and hours are subject to change without notice.

Sales and Non-Technical Customer Service

For product information, sales, and other non-technical

customer service, please email or call:

sales@bias-inc.com

US: 1-800-775-BIAS (2427) International: +1-707-782-1866

9AM-5PM Monday-Friday PST - excluding BIAS holidays

No technical support is available at the numbers or email address above.

Table of Contents

Chapter 1 - Introducing the Master Perfection Suite

Welcome!	.11
Minimum System Requirements	13
About Your User's Guide	13
Getting Help with the Master Perfection Suite	
Conclusion	.14

Chapter 2 - Installation, Registration, and Authorization

Installing the Master Perfection Suite	17
What the Master Perfection Suite Installs	17
Registration & Authorization	19
Alternate Methods of Registration/Authorization	20
Deauthorizing a Computer	22
Conclusion	22

Chapter 3 - GateEx

Introduction	25
GateEx User Interface & Controls	25
Global Controls	25
Gate/Expander Controls	27
Graphic Displays	29
Parameter Automation	30
Using GateEx	30
Conclusion	31

Chapter 4 - PitchCraft

Introduction	35
PitchCraft User Interface & Controls	35
Global Controls	35
Tuning History Graph Display & Tuning Meter	
Transposition Controls	
Pitch Correction Controls	39
Keyboard Controls	40
Custom Scale Controls	41
Parameter Automation	41
Using PitchCraft	41
Conclusion	43

Chapter 5 - Repli-Q

Introduction	.47
Repli-Q User Interface & Controls	.47
Global Controls	.47
Profile Controls	.49
Graph Display	.51
Parameter Automation	.53
Using Repli-Q	.53
Conclusion	.54

Chapter 6 - Reveal

Introduction	57
Reveal User Interface & Controls	57
Global Controls	57
Tool Specific Controls & Displays	60
Oscilloscope Tool	60
Peak & RMS Power History Tool	61
Spectrogram Tool	61
Pan Power/Phase Correlation Tool	62
Spectral Analysis Tool	62
Lissajous Phase Scope Tool	64
Peak & RMS Level Meters Tool	64
Parameter Automation	66
Conclusion	66

Chapter 7 - SuperFreq

ntroduction	69
SuperFreq User Interface & Controls	69
Global Controls	69
Gaulization Controls	72
requency Graph Display	73
arameter Automation	74
Ising SuperFreq.	74
Conclusion	75
7011c1d01011	

Sqweez-3 & -5

Introduction	79
Sqweez-3 & -5 User Interface & Controls	79
Global Controls	79
Master Compression Controls	81
Graph Display	82
Band-Specific Controls	83
Group Controls	84
Parameter Automation	84
Using Sqweez-3 & -5	85
Conclusion	85

Chapter 1: Introducing the Master Perfection Suite



Chapter 1: Introducing the Master Perfection Suite

Welcome!

Thank you for purchasing the BIAS Master Perfection Suite! This collection of powerful audio tools was designed in conjunction with professional audio engineers, based on their feedback and production needs. The Master Perfection Suite provides the ultimate tools for mixing, mastering, sound design, and analysis.

The Master Perfection Suite includes six professional audio plug-ins:

- GateEx Gate/Downward Expander
- PitchCraft Pitch Correction/ Transposition/Modification
- Repli-Q EQ Matching/Linear Phase EQ
- Reveal 7 Advanced Analysis Tools in One Intuitive Interface (Peak/RMS Level Meters, Lissajous Phase Scope, Oscilloscope, Pan Power Meter, Peak/RMS Power History, Spectral Analyzer, Spectrogram)
- Sqweez-3 & -5 Linear Phase Multiband Compressors/Expanders/Limiters
- SuperFreq-4, -6, -8 & -10 Parametric Equalizers

Master Perfection Suite Features (All Plug-Ins)

- Powerful, professional audio processing
- Convenience works with practically any Audio Units, RTAS/AudioSuite, or VST host on Mac OS X, Windows XP, and Windows Vista

• Easy-to-use and intuitive user interfaces with tools for practically any element of the mixing, mastering, or sound design process

GateEx Features

- User-Selectable Gate or Expander Functionality
- High-Precision Input/Output Level Meters
- Threshold Control
- Attack Control
- Release Control
- Depth Control
- Hold Control
- Hysteresis Control
- Lookahead Control
- Ratio Control
- Gain Control
- Waveform Display with Integrated Gate Threshold
- High-Precision Graph Display
- A/B/C/D Compare Snapshots
- Numerical Value Entry Fields

- Keyboard Parameter Nudging
- Parameter Automation (in Compatible Hosts)

PitchCraft Features

- User-selectable Reference Source
- User-selectable Pitch Range
- Pitch Transposition
- Formant Modification
- Automatic Formant Preservation
- User-selectable Pitch Correction Tuning
- User-selectable Pitch Correction Scale
- User-selectable Pitch Correction Key
- User-selectable Pitch Correction Time
- User-selectable Detuning
- Keyboard display with Current Scale Display
- User-Defined Custom Tuning Scale
- High-Precision Graph with Tuning History Display
- Tuning Meter
- A/B/C/D Compare Snapshots
- Numerical Value Entry Fields
- Keyboard Parameter Nudging
- Parameter Automation (in Compatible Hosts)

Repli-Q Features

- "Learn" Frequency Spectrum Function
- Save & Load Learned Frequency Spectrums
- User-selectable Reference Frequency Spectrum
- Save & Load EQ Function
- Spectrum Frequency Matching

- Spectrum Frequency Smoothing
- Gain Control
- Automatic Gain Compensation
- Soft Clip
- High-Precision Graph with Profile & EQ Spectrum Display
- A/B/C/D Compare Snapshots
- Numerical Value Entry Fields
- Keyboard Parameter Nudging
- Parameter Automation (in Compatible Hosts)

Reveal Features

- Peak/RMS Level Meters
- Lissajous Phase Scope
- Oscilloscope
- Pan Power Meter
- Phase Correlation Meter
- Peak/RMS Power History
- Spectral Analyzer
- Spectrogram
- User-Selectable Input Source
- Global Display Mode
- Individual Tool Display Mode
- User-Selectable Level Meter Range
- Global Freeze for all Displays
- A/B/C/D Compare Snapshots
- Numerical Value Entry Fields
- Keyboard Parameter Nudging
- Parameter Automation (in Compatible Hosts)

Sqweez-3 & -5 Features

- Includes two separate plug-ins (3 & 5-band editions)
- Automatic Gain Compensation
- Soft Clip
- User-Selectable Digital Ceiling
- Integrated Spectrograph Display
- High-Precision Input/Output Level Meters
- Gain Control Global & Per Band
- Threshold Control
- Ratio Control
- Maximum Reduction Control
- Attack Control
- Release Control
- User-Adjustable Knee
- Individual Band Bypass
- Individual Band Solo
- Ratio/Max Reduction/Attack/Release/ Knee Parameter Grouping
- High-Precision Graphic Display with EQ/ Threshold/Both View
- A/B/C/D Compare Snapshots
- Numerical Value Entry Fields
- Keyboard Parameter Nudging
- Parameter Automation (in Compatible Hosts)

SuperFreq Features

- Includes 4-, 6-, 8-, and 10-Band Editions
- Peak, High Shelf, Low Shelf, High Cut, and Low Cut Filters
- -18dB to + 18dB Gain Values

- .1 to 30 Q Values
- 20Hz-20kHz Frequency Values
- Individual Band Bypass
- -24 to +24 dB Logarithmic Output Level Control
- -24 dB Logarithmic Meters
- A/B/C/D Compare Snapshots
- Numerical Value Entry Fields
- Keyboard Parameter Nudging
- Parameter Automation (in Compatible Hosts)

Minimum System Requirements

Systems must meet the minimum requirements of the host application or platform when that is higher than the Master Perfection Suite minimum system requirements stated here.

To use the Master Perfection Suite on a Macintosh – you will need:

- G4, G5, or Intel-based Mac
- Mac OS X v.10.3.9, 10.4 (≥500 MHz processor), or 10.5 (≥867 MHz processor)
- $\geq 1 \text{ GHz processor recommended}$

To use Master Perfection Suite on a Windows PC - you will need:

- Intel P3/P4, Athlon w/SSE Instructions (≥800 MHz processor)
- ≥ 2 GHz processor recommended
- Windows XP Home/Professional with Service Pack 2 or Windows Vista

Both:

• Compatible Audio Units (Mac only), RTAS/AudioSuite, or VST host

- 256 MB RAM (512 MB RAM or higher recommended)
- 300 MB available hard disk space
- Color monitor, minimum 1024 x 768 resolution

For the most up-to-date info, please visit:

http://www.bias-inc.com/products/masterperfectionsuite/

About Your User's Guide

The Master Perfection Suite is designed to be powerful, yet intuitive. Your User's Guide is designed to help you set up and use the Suite for processing as quickly and easily as possible.

This User's Guide assumes that you are familiar with standard computer operating techniques, including:

- Setting up, starting, and using your computer
- Choosing commands from menus
- Clicking, double-clicking, selecting, shift-selecting, dragging with the mouse, and other common operations
- Opening, copying, saving and deleting files
- Opening, closing, scrolling, moving, re-sizing, and selecting windows

If you don't know how to perform these tasks, please refer to your computer's documentation and spend a little time learning about your computer before going any further. This will make working with the Master Perfection Suite much easier and more enjoyable.

The chapters in your Master Perfection Suite User's Guide are arranged in the order listed below, and will familiarize you with each plug-in's interface, controls, and parameters – and will give basic getting started tips for using them:

• Chapter 1 introduces you to the Master Perfection Suite and explains some of the requirements for using it

- Chapter 2 explains how to install, register, and activate the Master Perfection Suite
- Chapter 3 introduces you to GateEx
- Chapter 4 introduces you to PitchCraft
- Chapter 5 introduces you to Repli-Q
- Chapter 6 introduces you to Reveal
- Chapter 7 introduces you to SuperFreq
- Chapter 8 introduces you to Sqweez-3 & -5



Look for important tips and notes whenever you see this exclamation mark.

Getting Help with the Master Perfection Suite

This User's Guide will answer many common questions about each plug-in's controls. For additional help on usage and techniques, you may wish to access tutorials, technical support, and FAQs, all available via the BIAS website:

http://www.bias-inc.com

BIAS recommends using these resources before contacting Technical Support. The answers to most common technical questions can be found online. Best of all, using online help resources is free, easy, and available 24 hours a day, 7 days a week, 365 days a year. If you do need to contact Technical Support, please see the BIAS Technical Support Policy on page 5 of this user's guide.

Conclusion

Now that you know a little about the Master Perfection Suite, proceed to the next chapter to learn how to install your software and get started using it.

Chapter 2: Installation, Registration, and Authorization



The Master Perfection Suite's auto-installer softwaremakes installation very easy. The steps below will guideyou through the process.

Your complete Master Perfection Suite system consists of:

Installing the Master Perfection Suite

Installation, Registration,

and Authorization

CD-ROM installer

Chapter 2:

- Serial number
- Authorization File

Before you install the Master Perfection Suite, please check the Minimum System Requirements on Page 13.

If you purchased any of the Master Perfection Suite plug-ins separately, you may choose to install only for which you have a serial number. If you choose to install the entire suite, please be aware that plug-ins you do not have a serial number for will operate only in trial mode for a limited time.

To install the Master Perfection Suite on a Macintosh:

- 1. If you are using any virus protection software, turn it off or temporarily remove it, and restart your Macintosh.
- 2. Insert the Master Perfection Suite installer CD-ROM into your CD-ROM drive when it appears on your computer's desktop, double-click the BIAS Master Perfection Suite.mpkg icon.
- 3. The Installer will prompt you about running a program to determine if the Master Perfection Suite

can be installed - click Continue when this appears.

- 4. When the Installer dialog appears, click the Continue button.
- 5. Read the Software License Agreement then click the Continue button to proceed.
- 6. Click the Agree button if you agree to the terms of the Software License Agreement.
- 7. Select the hard drive/volume on which you wish to install the Master Perfection Suite, and click the Continue button.
- 8. Click the Install/Upgrade button (depending on the configuration of your user account, you may need to enter your account password. If prompted for your password, enter it and click the OK button. If you do not know your account password, please contact your system administrator for assistance).
- 9. When the installation is complete, a message will appear indicating that the installation was successful. Click the Close button to quit the Installer. (Don't forget to turn back on any virus-protection software that you may be using the next time you restart the computer).

To install the Master Perfection Suite on a Windows PC:

- 1. If you are using any virus-protection software, turn it off or temporarily remove it, and restart your computer.
- 2. Insert the Master Perfection Suite Installer CD-ROM in your CD-ROM drive. If the Master Perfection Suite Setup program does not automatically launch, double-click the Install Master Perfection Suite icon.

- 3. When the InstallShield Wizard for BIAS Master Perfection Suite appears, click the Next button.
- 4. Read the Software License Agreement, and then click the "I accept the terms in the license agreement" radio button to continue with installation.
- 5. To install the Master Perfection Suite into the default directory (recommended), click the Next button (Vista users skip this step).
- 6. In the Setup Type dialog, choose the Typical option (Vista users choose the Complete option), and click the Next button.
- 7. In the Select Program Folder dialog, click the Next button to continue with installation.
- 8. In the Ready to Install dialog, click the Next button to continue with installation.
- 9. When you are finished installing, click Finish to exit the installer.



At the end of installation, the Installer program will instruct you to authorize the plug-in – this is described later in this chapter.

What the Master Perfection Suite installs

The Master Perfection Suite installer automatically places various formats of the selected plug-ins, and documentation on your computer.

Mac OS X-based Systems:

The files and install locations for Mac OS X-based computers are described below:

Audio Units Hosts: (Peak, Digital Performer, GarageBand, Logic, etc.)

The installer automatically places the Audio Units format plug-ins into the following directory:

/Macintosh HD/Library/Audio/Plug-Ins/Components/

RTAS/AudioSuite Hosts (Pro Tools):

The installer automatically places the RTAS/AudioSuite format plug-ins into the following directory:

/Macintosh HD/Library/Application Support/ Digidesign/Plug-Ins/

VST Hosts: (Peak, Cubase, Live, Nuendo, etc.)

The installer automatically places the VST format plug-ins into the following directory:

/Macintosh HD/Library/Audio/Plug-Ins/VST/

Documentation

An electronic (PDF format) version of the Master Perfection Suite User's Guide, and a Read Me file with late-breaking information are both installed into:

/Macintosh HD/Library/Documentation/BIAS/Master Perfection Suite /



Check the documentation included with your bost application for more information on where it expects plug-ins to reside in order to function properly.

Windows XP & Vista-based Systems:

The files and install locations for Windows-based computers are described below:

RTAS/AudioSuite Hosts (Pro Tools):

The installer automatically places the RTAS/AudioSuite format plug-ins into:

/Program Files/Common/Digidesign/DAE/Plug-Ins/

2

VST Hosts: (Cubase SL/SX, Nuendo, etc.)

The installer automatically places the VST format plugins into the VstPlugIns folder of your VST host program.

Documentation

An electronic (PDF format) version of the Master Perfection Suite User's Guide, and a Read Me file with late-breaking information are both installed into:

/Program Files/BIAS/BIAS Master Perfection Suite/

If you would like to register and authorize the Master Perfection Suite at this point, simply attempt to use one of the component plug-ins – when an unauthorized copy is first launched, the Welcome to Master Perfection Suite dialog appears – click the Authorize button to proceed to the BIAS Authorization Manager. Have your Owner's Certificate handy, and follow the directions in the next section.

Registration & Authorization

The Master Perfection Suite must be authorized to work on your computer. You must register the Master Perfection Suite with BIAS in order to authorize it.

To register and authorize your software, you will need Internet access – either on the computer where the software will be used, or on another computer.

Serial Number-Based Authorization System

The Master Perfection Suite's authorization system consists of the following components:

- **BIAS Authorization Manager** A simple application for managing BIAS software licenses.
- Serial Number Located on Owner's Certificate, or on

the electronic receipt from a downloadable purchase.

- *Registration Account* Your registration and authorization information on the BIAS Server.
- **Authorization Request File** A file transferred from your computer to the BIAS server, which adds the computer you are authorizing to your BIAS Registration Account.
- **Authorization File** A file transferred from the BIAS server to your computer, which authorizes it to run the Master Perfection Suite.



You must register with BIAS in order to authorize your computer – the Master Perfection Suite will not launch unless you authorize it.

The registration/authorization process provides the following services:

- Registers your software with BIAS making you eligible for technical support and product updates.
- Authorizes your computer so it can run the Master Perfection Suite. You may authorize up to two computers for use with the Master Perfection Suite

 for example, a studio or office computer, and a laptop for home or mobile use.



To Register & Authorize a Computer to Run the Master Perfection Suite:

 Attempt to run one of the Master Perfection Suite plug-ins in your host program – the first time it is opened on an unauthorized computer, you will be prompted to use it in trial mode, or to authorize it – click the Authorize button.

2. When the BIAS Authorization Manager appears, select BIAS Master Perfection Suite 1.x in the "Please Choose a Product" section – then enter your Master Perfection Suite serial number (your serial number is printed on the Owner's Certificate included in the Master Perfection Suite package, or on the electronic receipt from a downloadable purchase) and email address in the appropriate fields, and click the Authorize button.



Your serial number appears on the Owner's Certificate, included in the Master Perfection Suite package, or on the electronic receipt from a downloadable purchase (not shown here).

If you bave purchased individual Master Perfection Suite plug-ins, be sure to select a specific plug-in in the step above, instead of the entire Master Perfection Suite. If you have purchased more than one, you will need to repeat the authorization process for each plug-in (plug-ins purchased separately will each have a unique serial number). Depending on how you access the BIAS Authorization Manager (i.e., launching the stand-alone version, or by instantiating an unauthorized Master Perfection Suite plug-in), you may need to click the "Show All BIAS Products" button, to have specific plug-ins appear in the Authorization Manager's product list.



If you use more than one email address, be sure to make a note of which one you registered with – this will make managing your BIAS Registration Account faster and simpler in the future.

- 3a. If you have previously registered another BIAS product, using the same email address, the Master Perfection Suite (or individual plug-in) will be authorized automatically – please continue to step 4.
- 3b. If the BIAS Registration system does not recognize your email address (i.e., if this is the first BIAS product you are registering, or if you enter a different email address than with previously registered BIAS products) – a dialog will appear, prompting you to enter your registration information. Enter your registration information, and click the Submit button – a dialog will appear indicating that you have been successfully registered – please continue to step 4.
- 4. A dialog will appear indicating successful authorization click the OK button, and then click the Done button to exit the Authorization Manager, and begin using the Master Perfection Suite.

Alternate Methods of Registration/Authorization

If the computer on which you plan to use the Master Perfection Suite cannot be temporarily connected to the internet, even for this one-time authorization process, there are a number of other ways to register and authorize the Master Perfection Suite.

Registering & authorizing the Master Perfection Suite using another computer with internet access

To use this method of registration/authorization, you will need to run the BIAS Authorization Manager program

from your installer CD-ROM – or copy the BIAS Authorization Manager from your installer CD-ROM to the internet-equipped computer's hard drive and run it from there.

To Register & authorize using another computer with internet access:

- Attempt to open the Master Perfection Suite on the computer where it will be used (i.e., the computer without internet access) – the first time one of the Master Perfection Suite plug-ins is used on an unauthorized computer, the Welcome to Master Perfection Suite dialog appears – click the Authorize button to proceed to the BIAS Authorization Manager.
- 2. In the BIAS Authorization Manager dialog, select BIAS Master Perfection Suite 1.x in the "Please Choose a Product" section – then enter your Master Perfection Suite serial number and email address (your serial number is printed on the Owner's Certificate included in the Master Perfection Suite package, or on the electronic receipt from a downloadable purchase).

If you have purchased individual Master Perfection Suite plug-ins, be sure to select a specific plug-in in the step above, instead of the entire Master Perfection Suite. If you have purchased more than one, you will need to repeat the authorization process for each plug-in (plug-ins purchased separately will each have a unique serial number). Depending on how you access the BIAS Authorization Manager (i.e., launching the stand-alone version, or by instantiating an unauthorized Master Perfection Suite plug-in), you may need to click the "Show All BIAS Products" button, to have specific plug-ins appear in the Authorization Manager's product list.

- Click the Alternate Authorization button the Alternate Authorization dialog appears.
- 4. In the Alternate Authorization dialog, click the Generate Authorization Request button the

Serial Number/Email Address dialog appears.

- 5. Confirm that the email address and serial number are correct, and click the OK button to save an Authorization Request File to your hard drive.
- 6. Transfer the Authorization Request File to another computer that has internet access (via a LAN, by burning it to CD, copying to a floppy diskette, flash drive, external hard drive, iPod, digital camera, etc.)
- 7. Launch the BIAS Authorization Manager program on your internet-equipped computer.
- 8. In the "Please Choose a Product" section, choose BIAS Master Perfection Suite 1.x (or individual plug-in if purchased separately).
- 9. Click the Alternate Authorization button in the BIAS Authorization Manager program.
- 10. Go directly to step 3 in the Alternate Authorization dialog click the Open Authorization Request button.
- In the Open dialog that appears, locate and select the Authorization Request File, then click the Open button – this will transmit the Authorization Request File to the BIAS Server.
- 12a. If you have previously registered another BIAS product, using the same email address, the BIAS server will transmit an Authorization File back to your internet-equipped computer click the Save button to save this to the internet-equipped computer's hard drive. Please proceed to step 13.
- 12b. If the BIAS Registration system does not recognize your email address (i.e., if this is the first BIAS product you are registering, or if you enter a different email address than with previously registered BIAS products) – a dialog will appear, prompting you to enter your registration information.
- 12c. Enter your registration information, and click the Submit button, then click the OK button when the successful registration dialog appears.

- 12d. You will be prompted to open the Authorization Request File a second time – locate it, and then click the Open button – The BIAS server will transmit an Authorization File back to your internet-equipped computer – click the Save button to save this to the internet-equipped computer's hard drive.
- 13. Click OK to exit the Authorization Successful dialog, then click the Done button in the Alternate Authorization dialog, and again in the BIAS Authorization Manager.
- 14. Transfer the Authorization File (the file is called "BIAS Master Perfection Suite 1.x" for the whole suite, or "BIAS < Plug-in Name > 1.x" for individual plug-ins) from the internet-equipped computer back to the host computer where the plug-in(s) will be used (via a LAN, by burning it to CD, etc.).
- 15. Go directly to step 4 in the Alternate Authorization dialog click the Open Authorization File button.
- 16. In the Open dialog that appears, locate and select the Authorization File, then click the Open button.
- 17. Click OK in the Authorization Complete dialog, and Done in the Authorization Manager.

To register & authorize the Master Perfection Suite with no internet access

If you do not have temporary internet access on any computer, or need help with any part of the registration/authorization process, please contact BIAS at:

US Toll-Free: 1-800-775-2427 International: +1-707-782-1866

Deauthorizing a Computer

If you need to de-authorize a particular computer that will no longer be used to run the Master Perfection Suite, you can do so by following the steps below.



The computer running the Master Perfection Suite must be connected to the internet in order to be deauthorized.

To De-Authorize a Computer:

- 1. Launch the BIAS Authorization Manager from your hard drive (or launch the BIAS Authorization Manager program from your installer CD-ROM).
- 2. In the BIAS Authorization Manager, select the product you wish to de-authorize the serial number and email address used to authorize that product will be automatically recalled and will appear in the Serial Number and Email Address fields.
- 3. Click the Deauthorize button.
- Click the Deauthorize button to confirm the deauthorization process – the Deauthorization Successful dialog appears.
- 5. Click the OK button then click the Done button in the BIAS Authorization Manager.

Conclusion

Now that you have installed, registered, and authorized the Master Perfection Suite, please proceed to the next chapter to learn about using it.

Chapter 3: GateEx



Chapter 3: GateEx

Introduction

GateEx is a full-featured noise gate and downward expander plug-in. When used as a noise gate, GateEx allows an audio signal to be reduced in amplitude when it falls below a user-defined threshold level. When used as a downward expander, GateEx allows the apparent dynamic range of an audio signal to be increased.

Some common examples of how a gate would typically be used are:

- In dialogue recordings, a gate might be used to remove (or minimize) the presence of background noise or room tone that is audible in the amplitude dips between the speaker's words or phrases. When the subject speaks, the rising amplitude of the voice triggers the gate to open, allowing the voice to pass through, and as words trail off and the amplitude drops, the gate closes and prevents any low level background noise from being audible.
- In music recordings (we'll use a drum kit as an example), a gate might be used to remove or isolate one type of drum from another. When used properly on a drum recording, a gate can accomplish such tasks as isolating the snare drum, and removing any bleed picked up through the snare mic from other nearby drums, such as the kick or hi-hat.

A common example of how a downward expander would typically be used:

• A downward expander can be applied to increase the apparent dynamic range. This increase in

dynamic range is accomplished by decreasing the gain amount in the relatively quiet portions of the recording, so that the apparent noise floor is lowered.

GateEx User Interface & Controls

The GateEx plug-in is designed to be intuitive and easy to use. This section discusses the graphical user interface, and will familiarize you with each of the various controls.

Global Controls

GateEx features a variety of parameters designed for specific tasks, and also features some "global" controls, that affect the state of the entire plug-in. Below is a brief description of these "global" tools, and how they operate.

A/B/C/D Compare Buttons

The A/B comparison feature is common in audio programs, and gives the ability to preview different settings before committing to them. Essentially, various tool settings are made and stored as snapshot "A" -

GateEx Graphical User Interface



*Appearance of preset/bypass controls will vary depending on the platform, host, and plug-in format used

another group of settings can then be assigned as snapshot "B" – and then clicking the A or B buttons in the interface instantly toggles between the two. As GateEx offers many fine-tuning options, it features not just two comparison modes – but instead includes A/B/C/D comparisons, for four unique snapshots to compare between, before committing to any file processing. These snapshots are similar to saving and loading presets, but are instantaneous, and do not require selecting presets from a menu to load them. To permanently save one of these snapshots, simply select the desired snapshot, and save it as a custom User Preset.

Factory Preset Pop-Up Menu

This menu displays the current factory preset, if one has been selected. Clicking this menu shows a list of all available factory presets. A factory preset may be selected by choosing it from this pop-up menu.

Factory Preset Selection Arrows

These arrows are used to navigate through factory presets. Clicking the down arrow moves to the next available factory preset, and clicking the up arrow moves to the previous factory preset.

User Preset Save Button

Saves a user preset, which includes the state of all GateEx parameters.

User Preset Load Button

Recalls a user preset, includes the state of all GateEx parameters.



User presets do not appear in the Factory Preset pop-up menu, so be sure you are aware of the directory where you are saving them into, so they may be recalled easily later.

Bypass Button

The Bypass button is used to temporarily disable the plug-in. When the plug-in is bypassed, audio playback is not affected by the plug-in.

Gate/Expander Controls

The Gate/Expander controls each affect a unique characteristic of the overall gate/expander behavior. These are described briefly below.

Input Level Meters

The Input Level Meters show the level of the original audio signal being fed into the plug-in, before it is affected by any of the various controls.

Output Level Meters

The Output Level Meters show the level of the processed audio signal, after being affected by the active parameter settings.

Threshold Slider

The Threshold slider determines the amplitude level at which the gate opens or closes. If the signal level exceeds the Threshold setting, the gate opens, and remains open for as long as the signal remains above the Threshold. If the signal falls below the Threshold setting, the gate closes, and remains closed until the signal exceeds the Threshold setting.

By default, the Threshold slider is set to a value of -60dB, and has a range of 0 to -60dB.

Attack Knob

The Attack Knob determines how quickly the gate opens when the audio signal being fed into it exceeds the threshold level.

By default, the Attack Knob is set to a value of 30ms, and has a range of 1-1000ms.

Release Knob

The Release Knob determines how quickly the gate closes when the audio signal being fed into it falls below the threshold level.

By default, the Release Knob is set to a value of 100ms, and has a range of 50-3000ms.

Depth Knob

The Depth Knob determines how far the gate closes (i.e., how deeply the signal is attenuated) when triggered by a signal that exceeds the threshold level. By default, the Depth control is set to its maximum level, -90dB, which causes audio signals to be completely silenced when the gate is in the closed position. The Depth control is useful as it allows a variable level of signal reduction when the gate is closed, and can help produce a more subtle gating effect. Another useful effect of having a Depth setting lower than the maximum amount is that it allows the gate to open faster than if it were in the fully closed position, and can be useful if a faster attack time is needed, and the Attack knob is already set to operate at its lowest value (1ms).

By default, the Attack Knob is set to a value of -90dB, and has a range of 0 to -90dB.

Hold Knob

The Hold Knob allows the gate to remain in the open position for a user-specified amount of time, even after the signal level has fallen below the threshold level. This control is especially useful when the audio signal is very close to the threshold level, and might cause the gate to open and close repeatedly, very quickly, as the signal exceeds and then falls below the threshold over a very short period of time. This rapid opening and closing of the gate can create an unwanted artifact called "chatter" or "jitter", and the Hold control helps to avoid it.

By default, the Hold Knob is set to 0ms, and has a range of 0 - 3000 ms.

Hysteresis Knob

The Hysteresis control is similar to the Hold control in its practical application of helping to avoid/eliminate "chatter", but achieves this in a different way. Rather than simply maintaining the gate's open position for a user-specified duration, the Hysteresis control allows setting independent threshold levels for opening or closing the gate, by specifying the difference between them in decibels. With this approach, the gate opens when a primary threshold level is exceeded by a suitable signal level, and closes again only when the signal level drops below a secondary threshold level.

By default, the Hysteresis Knob is set to 0dB, and has a range of 0 - 10dB.

Lookahead Knob

As its name implies, the Lookahead Knob reads data from a file a bit earlier than it is actually played back. This allows the gate to react earlier than it would, based on its threshold setting alone. This feature is useful as it helps to preserve transients and attacks that might be eliminated (or minimized) due to the gate opening after they had already occurred.

By default, the Lookahead Knob is set to 0ms, and has a range of 0 - 100ms.

Ratio Knob

The Ratio Knob is used to set the rate at which audio signal reduction is applied, when the signal exceeds the threshold level.

By default, the Ratio Knob is set at 1:1, and has a range of 1:1 - 50:1.

Gain Slider

The Gain slider is used to set the output level. By default, it is set to 0dB, and has a range of 0 to -90dB.



Sliders & knobs can be reset to their default values by Alt/Option-clicking on them.



(i.e., after clicking in them with the mouse), their values may be changed by typing in new values, by using the Up/Down Arrow keys on your keyboard, or by using the scroll wheel on a compatible mouse.

When numerical fields are in an editable state

Graphic Displays

GateEx features two graphic display areas that are helpful in making the appropriate settings quickly. These are described below.

Waveform Display

The Waveform Display shows a graphical representation of the audio signal being fed into GateEx in real-time. This display is particularly helpful in quickly setting the threshold level, as it creates a graphical relationship between the signal level and the threshold level.

Gate Threshold Indicators

These indicators appear as horizontal white lines superimposed over the Waveform Display, and give a graphical representation of the current threshold value, in relation to the input signal level.

Freeze Waveform Button

The Freeze Waveform Button allows you to stop and start the waveform from scrolling. This is useful for examining particular areas of the waveform display as you scroll through its timeline.

Waveform View Radio Buttons

This group of radio buttons allows choosing between having the Waveform Display calculated by the content of a stereo signal's left channel (L), right channel (R), or both channels (L+R).



When using L+R mode, a single waveform displays the sum of both channels.

Zoom Button

This control is located where the X & Y-axes meet. It can be used to zoom in/out on either axis independently, or both at the same time. This allows for higher precision viewing of the graph display.

Horizontal Scrollbars/Arrows

When zoomed in on the horizontal axis, the scrollbars and/or scroll arrows may be used to move the Waveform Display earlier or later in time, allowing examination of waveform events anywhere along the timeline.

By default, the zoom level is set to "zoomed out all the way", and the horizontal scrollbars/arrows have no effect.

Vertical Scrollbars/Arrows

When zoomed in on the vertical axis, the scrollbars and/or scroll arrows may be used to move the Waveform Display up or down, allowing examination of events that occur in the waveform's positive or negative phase. By default, the zoom level is set to "zoomed out all the way", and the vertical scrollbars/arrows have no effect.

Graph Display

The Graph Display shows a number of key parameter settings in an easy-to-read graph format, with each axis labeled in decibels. The X-axis represents audio input level, while the Y-axis represents output level. The graph makes it fast and easy to make settings, and allows easy plotting of input and output level values.



Parameter Automation

GateEx's various controls may be automated (i.e. applied dynamically) within hosts that support parameter automation. For more information, please consult your host's documentation on how to automate parameters.

Using GateEx

This section contains some basic "hands-on" getting started tips.

1. Use the Ratio Knob to set the rate at which reduction increases when the audio signal exceeds the current threshold level. A good starting point is to set it to the highest setting (i.e., 50:1).



2. Adjust the Threshold slider until the gate opens to allow the desired audio content to pass through it, and closes to prevent lower level, unwanted audio from being passed.



- 3. Fine-tune your settings by making adjustments to the following controls, and try these tips:
- **Attack and Release** Use faster times for percussive material (such as drums), or slower times for less-percussive material (such as voice).
- **Depth** Set the depth so the proper amount of reduction is happening when the gate is closed. This allows for subtle gating where not all the signal is removed when the gate is closed, or more drastic gating where all the signal is removed when the gate is closed.
- *Hold* Use the hold control to ensure the gate stays open for the desired length of time (i.e., so that it does not begin to close prematurely and reduce part of the desired audio signal).
- *Hysteresis* Use the hysteresis control to set independent threshold levels for opening/closing the gate if the audio signal hovers near the threshold setting.
- **Lookahead** Use the Lookahead control if you need the gate to open a little before the audio signal goes above the threshold level in order to preserve any transients or attacks.

If you are unfamiliar with using gates, try using some of the included factory presets, which are representative of common gate settings. These should provide a good starting point, from which you can fine-tune the controls to best suit the audio material you are working with.



Conclusion

Now that you are familiar with the various GateEx controls, and its basic workflow, please continue along to the next chapter, where you will learn about the PitchCraft plug-in.

Chapter 4: PitchCraft



Chapter 4: PitchCraft

Introduction

PitchCraft is a pitch correction and transposition plug-in that can be used to correct and/or change pitch from a variety of sources. The high quality algorithm includes formant correction to retain the natural sound of a voice – and may also be used to intentionally modify the characteristics of a voice.

This plug-in provides high-quality professional pitch correction/transposition, with an easy to use, intuitive interface.

Some common examples of how PitchCraft might be used include:

- To correct pitch problems in vocal or instrumental recordings
- Change pitch of a performance to match a different key than was used in the original performance
- Pitch and formant modification, as a sound design tool
- Virtually altering the size of the vocal cavity and vocal cords, for effects such as "gender transformation", voice disguising, changing an adult's voice into a child's voice and vice-versa

PitchCraft User Interface & Controls

The PitchCraft plug-in is designed to be intuitive and easy to use. This section discusses the graphical user interface, and will familiarize you with each of the various controls. The user interface is divided into a number of sections, each with dedicated controls or displays used for pitch transposition and correction.

Global Controls

PitchCraft features a variety of parameters designed for specific tasks, and also features some "global" controls, that affect the state of the entire plug-in. Below is a brief description of these "global" tools, and how they operate.

Reference Source Pop-Up Menu

This menu allows you to choose whether the reference source (i.e., the signal being fed into PitchCraft) is a single (left or right) channel of a stereo file, or both channels simultaneously.



If you are working with a mono file, be sure to set the Reference Source Pop-Up menu to "Left".

Pitch Range Pop-Up Menu

This menu allows you to choose the desired pitch range to use. Using the appropriate pitch range optimizes PitchCraft for certain source types. The plug-in will not detect pitch values outside the bounds specified for a given source. If you hear artifacts in the audio output, 4

PitchCraft Graphical User Interface



*Appearance of preset/bypass controls will vary depending on the platform, host, and plug-in format used
you might want to try a different setting, as artifacts are generally produced when the source pitch falls outside the bounds specified by the value selected in the Pitch Range pop-up menu.

Making a pitch range selection in the Pitch Range pop-up menu assigns a corresponding pitch range scale along the Y-axis of the Tuning History Graph. Pitch range options include:

- Normal D2 (73 Hz) to E6 (1318 Hz)
- Female E3 (165 Hz) to E6 (1318 Hz)
- Male D2 (73 Hz) to B4 (494 Hz)
- Lower Register E0 (20 Hz) to E4 (330 Hz)
- Higher Register E4 (330 Hz) to D#7 (2500 Hz)

A/B/C/D Compare Buttons

The A/B comparison feature is common in audio programs, and gives the ability to preview different settings before committing to them. Essentially, various tool settings are made and stored as snapshot "A" another group of settings can then be assigned as snapshot "B" - and then clicking the A or B buttons in the interface instantly toggles between the two. As PitchCraft offers many fine-tuning options, it features not just two comparison modes - but instead includes A/B/C/D comparisons, for four unique snapshots to compare between, before committing to any file processing. These snapshots are similar to saving and loading presets, but are instantaneous, and do not require selecting presets from a menu to load them. To permanently save one of these snapshots, simply select the desired snapshot, and save it as a custom User Preset.

Factory Preset Pop-Up Menu

This menu displays the current factory preset, if one has been selected. Clicking this menu shows a list of all available factory presets. A factory preset may be selected by choosing it from this pop-up menu.

Factory Preset Selection Arrows

These arrows are used to navigate through factory presets. Clicking the down arrow moves to the next available factory preset, and clicking the up arrow moves to the previous factory preset.

User Preset Save Button

Saves a user preset, which includes the state of all PitchCraft parameters.

User Preset Load Button

Recalls a user preset, includes the state of all PitchCraft parameters.



User presets do not appear in the Factory Preset pop-up menu, so be sure you are aware of the directory where you are saving them into, so they may be recalled easily later.

Bypass Button

The Bypass button is used to temporarily disable the plug-in. When the plug-in is bypassed, audio playback is not affected by the plug-in.

Tuning History Graph Display and Tuning Meter

These controls are used to view an audio signal's tuning history over time, and also to customize and navigate the Graph display.

View Notes/Hz Radio Buttons

These controls allow you to choose whether the Graph display's Y-axis is represented as Notes or Hz (frequencies).

Graph Display

This display plots the pitch of both the input (raw) and output (processed) signals, and is useful for comparing the difference between the two. The yellow line represents the input pitch, and the orange line represents the output, or corrected pitch. The X-axis represents time, and the Y-axis represents the currently selected scale, and may be viewed as notes or Hz (frequencies).

Zoom Button

Zooming in or out is accomplished using the Zoom Button, located at the point where the X and Y-axes meet. This control allows a very detailed zoomed-in view, or can be zoomed-out for a "big picture" view. Option-clicking on the Zoom Button resets the zoom level.

- Zoom In Horizontally Click and drag to the right
- Zoom Out Horizontally Click and drag to the left
- Zoom In Vertically Click and drag up
- Zoom Out Vertically Click and drag down

Tuning Meter

The Tuning meter appears vertically along the right side of the interface, and displays the current amount of tuning. The divisions along the Tuning meter's scale is marked at 200-cent intervals, with the maximum tuning amount of 200 cents at the top of the scale, zero cents in the middle, and -200 cents at the bottom.

Transposition Controls

This group of controls is used for altering the pitch and formant of the reference source being fed into PitchCraft.

Transpose On/Off Checkbox

This control allows enabling/disabling all Transposition controls.

Pitch Slider/Numerical Field

This control allows raising or lowering the pitch value (in cents), while maintaining the original formant structure (i.e., the "timbre" or "tonal color" of the voice).



When working with vocal sources this control is the equivalent to modifying the length of the vocal cords.

Each division along the Pitch Slider's scale represents the associated musical interval (major second, major third, etc).



Sliders can be reset to their default values by *Alt/Option-clicking on them.*



When numerical fields are in an editable state (i.e., after clicking in them with the mouse), their values may be changed by typing in new values, by using the Up/Down Arrow keys on your keyboard, or by using the scroll wheel on a compatible mouse.

Formant Slider/Numerical Field

This control allows raising or lowering the formant value

4

(in cents) through a tilt of the spectral envelope, stretching the spectrum at the high end, and compressing it at the low end. The formant slider acts as a particular type of filter, while the pitch control performs the usual pitch transposition on the instrument.



When working with vocal sources this control is the equivalent to modifying the volume of the vocal cavity.

Each division along the Formant Slider's scale represents the associated musical interval (major second, major third, etc). When the Formant slider is set to zero, formants are preserved independently of pitch transposition (known as "formant-corrected pitch transposition").

Pitch Correction Controls

This group of controls is used for correcting problems with the pitch of the reference source being fed into PitchCraft.

Correct Pitch On/Off checkbox

This control allows enabling/disabling all Pitch Correction controls.

Tuning Pop-Up Menu

This menu allows you to choose the desired type of tuning to be used in the pitch correction process. Tuning is expressed in number of cents with respect to the equal-tempered note, for each of the twelve notes, and includes the following options:

• Equal – 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0

- Classic Just 0, -29, 4, 16, -14, -2, -10, 2, 14, -16, 18, -12
- Harmonic 0, 5, 4, -2, -14, -29, -49, 2, 41, 6, -31, -12
- Indian Raga 0, -10, 4, -6, -16, -2, -12, 2, -8, -18, -4, -14
- Arabic 0, 30, -20, -50, -45, 2, 23, 6, -14, -43, -70, 10
- Bali/Java-0, 27, -58, 42, -123, -130, -126, -211, 24, -61, -57, -142
- Tibetan 0, -42, 32, 10, -22, 22, 18, 25, -27, -4, 19, -14
- Carlos Alpha 0, -12, 44, 66, -88, -110, -132, -164, -176, -198, -220, -242
- Pythagorean 0, -10, 4, -6, 8, -2, 12, 2, -8, 6, -4, 10
- Mean 0, -24, -7, 10, -14, 3, -21, -3, -27, -10, 7, -17

Scale Pop-Up Menu

This menu allows you to choose the desired scale to be used in the pitch correction process. When a scale is chosen in the pop-up menu, PitchCraft's keyboard will highlight the notes used in that scale. The following scales are available:

- Chromatic
- Major
- Dorian
- Phrygian
- Lydian
- Mixolydian
- Minor
- Locrian
- Whole Tone
- Blues
- Harmonic Minor
- Melodic Minor
- Major Pentatonic
- Minor Pentatonic

- Major Bebop
- Dominant Bebop
- Super Locrian
- Diminished
- Hungarian Minor
- Arabian



PitchCraft's keyboard showing a "C" Dorian scale's notes.

Key Pop-Up Menu

This menu allows you to choose the root note of the scale selected in the Scale pop-up menu. Options include:

- A
- A#
- B
- C
- C#
- D
- D#
- E

- F
- F#
- G
- G#

Pitch Correction Time Slider

This control allows you to adjust the speed at which pitch correction is applied.

Detune Slider

This control allows you to adjust the reference frequency used for tuning (i.e., from 440Hz to 445Hz). This control should be adjusted only if attempting to match the pitch with another source that is not in "concert pitch" (i.e., not tuned to A 440).

Keyboard Controls

The Keyboard controls are used for defining custom notes and scales to be used in the pitch correction process. These controls are described below.

Ignore Note Checkboxes

These checkboxes allow certain notes to be ignored in the pitch correction process, even when they are present in the scale being used. The presence of a check in an Ignore checkbox means that the note represented by the key that appears directly below it in the interface will be ignored. The absence of a check means the note will be used in the pitch correction process.

Keyboard Display

The virtual keyboard is used to display the active notes

dynamically) within hosts that support parameter automation. For more information, please consult your

Parameter Automation

Using PitchCraft

This section includes hands-on directions for accomplishing common pitch modification tasks.

PitchCraft's various controls may be automated (i.e. applied

host's documentation on how to automate parameters.

To Define a Scale:

• Choose the desired Scale Preset from the Scale pop-up menu.

To Define a Custom Scale:

 Click the desired keys in PitchCraft's virtual keyboard – clicking a key toggles the Use Note in Scale indicators into the On or Off position.

Or:

- 1b. Click the keys of a MIDI input device (such as Peak's virtual MIDI keyboard) – the MIDI Note On messages toggle the Use Note in Scale indicators into the On or Off position.
- 2. When you have finished defining your custom scale, click the Done button, and enter a preset name the custom scale is added to the Scale menu as a preset, and becomes active.

PitchCraft's A/B/C/D snapshot controls will not store custom scales until they have first been saved into the Scale pop-up menu, as described in Step 2 in the previous section.

Custom Scale Controls

that note is not part of the current scale.

Use Note in Scale Indicators

These controls allow the creation of a custom scale to be used for pitch correction. Custom scales are automatically saved into the Scale pop-up menu, and may be recalled for use later.

being used in the pitch correction process, and can also be used to define a custom scale for pitch correction.

The Use Note in Scale indicators appear as orange dots superimposed over keys in the Keyboard display. Scales can be created or modified using any MIDI device, such as Peak's MIDI keyboard window, or a MIDI controller/keyboard. Any MIDI "Note On" message from a controller or sequence is added to the current scale, if the note is already part of the

current scale, the "Note On" message will remove it from the current scale. When a Use Note in Scale indicator appears

over a key, that key's note is detected and its pitch is

corrected. When an indicator does not appear over a key,

Done Button

Clicking the Done button activates a custom scale, and loads this scale's name into the Scale pop-up menu.

Reset Button

Clicking the Reset button clears current custom scale information, and resets the scale to that selected in the Key pop-up menu.

Current Scale Display

Displays the name of the current scale, as well as appropriately placed Use Note in Scale indicators over the keys/notes in the active scale.



To Correct Pitch:

- 1. Select the appropriate type of reference source from the Reference Source pop-up menu.
- 2. Select the appropriate pitch range for the reference source, using the Pitch Range pop-up menu.
- 3. Enable the Correct Pitch section, by checking the Correct Pitch checkbox.
- 4. Define a scale or set of notes using the Scale, Key, and Keyboard controls.
- 5. Play the audio source. As audio enters the plug-in and is analyzed, the current pitch is displayed in the Tuning History Graph. The plug-in determines which note in the current scale is "correct" and applies the appropriate amount of pitch correction to the input source, to match the note in the scale.

The Pitch Correction Time control can be used to smooth out the correction effect if the pitch becomes "warbly". It can also be used to create interesting "robotic" voice effects by setting the pitch correction time to zero.

The Transpose and Correct Pitch sections may be enabled simultaneously, with transposition and formant modification being performed in parallel with pitch correction. Advanced features include the ability to detune the reference frequency (for example from A = 440Hz to 443Hz), the ability to define the scale via a MIDI controller, and the ability to save up to 4 temporary presets.



PitchCraft is designed for monopbonic, harmonic sources and it will not work on polyphonic material – such as drums, cymbals, or mixes consisting of several instruments/voices. Please keep in mind that material with excessive reverb or background noise is not considered monophonic. It is advised that you clean up the audio signal as much as possible before processing it with this algorithm. PitchCraft may produce interesting creative effects when used with polyphonic audio material, but keep in mind this



is an algorithm carefully designed for transparent processing of monophonic sounds. For best results, the narrowest pitch range should be selected to match the input source's pitch range.

To Transpose Pitch and/or Formant:

- 1. Select the appropriate type of reference source from the Reference Source pop-up menu.
- 2. Select the appropriate pitch range for the reference source, using the Pitch Range pop-up menu.
- 3. Enable the Transpose section of the interface, by checking the Transpose checkbox.
- Adjust the Pitch and/or Formant sliders until the desired effect is achieved.

Voice Modification Using the Pitch & Formant Controls

The Pitch and Formant controls can produce very interesting special effects, especially when the source is a recording of a human voice. When using these controls for modifying a human voice, the following guidelines may be useful:

- Transposing pitch upward is the equivalent of shortening vocal cords
- Transposing pitch downward is the equivalent of lengthening vocal cords
- Shifting formants upward is the equivalent of reducing the size of the vocal cavity
- Shifting formants downward is the equivalent of enlarging the size of the vocal cavity

Some examples of voice modification possibilities are listed below:

To Transform a Female Voice into a Male Voice:

1. Using the Pitch slider, transpose pitch downward substantially.

2. Move Formant slider down until a natural sound is achieved, (which usually occurs when the Formant slider's value is slightly smaller than the Pitch slider's value).

To Transform a Male Voice into a Female Voice:

- 1. Using the Pitch slider, transpose pitch upward.
- 2. Move the Formant slider up until a natural sound is achieved.

To Transform an Adult's Voice into a Child's Voice:

• Raise the Formant slider's value – notice how "age" decreases as the formant value increases.

To Transform a Child's Voice into an Adult's Voice:

• Lower the Formant slider's value – notice how "age" increases as the formant value decreases.

To Transform a Human Voice into an "Alien" Voice:

- 1. Raise the Pitch value all the way (i.e., slider all the way to the right).
- 2. Lower the Formant value all the way (i.e., slider all the way to the left).

To Disguise a Human Voice:

- 1. Lower the Formant value slightly.
- 2. Adjust Pitch value to suit level of voice disguise.

Conclusion

Now that you are familiar with the various PitchCraft controls, and its basic workflow, please continue along to the next chapter, where you will learn about the Repli-Q plug-in.

Chapter 5: Repli-Q



Chapter 5: Repli-Q

Introduction

Repli-Q is a powerful spectral matching/balancing plugin, which is designed to analyze, edit, and compare the spectral content of audio recordings. Once analysis has taken place, spectral characteristics of one recording may be applied to another. By viewing the frequency spectrum plotted as graphical data (and by listening of course), a user can discern where there are peaks, "holes", or other characteristics in the spectral content of the recording. Various steps in the recording and mastering process can result in problems such as studio resonance caused by the natural comb filtering effect of a poorly shaped recording/listening environment - or equa-lization used to compensate for a studio's equipment and environmental deficiencies. Repli-Q allows these sorts of spectral problems to be located and repaired using its high quality linear-phase filters.

Some common examples of how Repli-Q might be used are:

- Mimicking and applying a favorite artist's spectral "style" to your own recordings.
- In the mastering process, it can give a cohesive sound to a collection of songs that may have been recorded and mixed in different locations/ environments.
- It can be used to create a collection of profiles microphones, for example. These profiles may later be applied to audio recordings to give them the characteristic sound of a particular type of microphone.

Repli-Q matching profiles may be created for other types of equipment, such as consoles, compressors, recording environments, etc.

Repli-Q User Interface & Controls

Repli-Q is designed to be intuitive and easy to use. This section discusses the graphical user interface, and will familiarize you with each of the various controls.

Global Controls

Repli-Q features a variety of controls designed for specific tasks, and also features a number of "global" controls, that affect the state of the entire plug-in. Below is a brief description of these "global" tools, and how they operate.

A/B/C/D Compare Buttons

The A/B comparison feature is common in audio programs, and gives the ability to preview different settings before committing to them. Essentially, various tool settings are made and stored as snapshot "A" -

Repli-Q Graphical User Interface



*Appearance of preset/bypass controls will vary depending on the platform, host, and plug-in format used

another group of settings can then be assigned as snapshot "B" – and then clicking the A or B buttons in the interface instantly toggles between the two. As Repli-Q offers many fine-tuning options, it features not just two comparison modes – but instead includes A/B/C/D comparisons, for four unique snapshots to compare between, before committing to any file processing. These snapshots are similar to saving and loading presets, but are instantaneous, and do not require selecting presets from a menu to load them. To permanently save one of these snapshots, simply select the desired snapshot, and save it as a custom User Preset.

Factory Preset Pop-Up Menu

This menu displays the current factory preset, if one has been selected. Clicking this menu shows a list of all available factory presets. A factory preset may be selected by choosing it from this pop-up menu.

Factory Preset Selection Arrows

These arrows are used to navigate through factory presets. Clicking the down arrow moves to the next available factory preset, and clicking the up arrow moves to the previous factory preset.

User Preset Save Button

Saves a user preset, which includes the state of all Repli-Q parameters.

User Preset Load Button

Recalls a user preset, which includes the state of all Repli-Q parameters.



User presets do not appear in the Factory Preset pop-up menu, so be sure you are aware of the directory where you are saving them into, so they may be recalled easily later.

Bypass Button

The Bypass button is used to temporarily disable the plug-in. When the plug-in is bypassed, audio is not affected.

Profile Controls

The Profile controls are used for learning, saving, and loading spectral profiles and EQ curves, as well as for matching and smoothing between different profiles. This section discusses each control's function.

Learn Spectrum Button

This control analyzes the spectral content of the audio source being fed into Repli-Q. When the Learn Spectrum button is clicked, Repli-Q immediately begins analyzing, or "learning" the input signal.

> When audio playback is stopped, the host program may still be sending digital silence (a zero amplitude signal for all samples) to the plug-in. If the Learn Spectrum button has been clicked, Repli-Q is learning the spectrum of digital silence – which is a flat spectrum of very low amplitude (-96 dB). If playback starts, Repli-Q continues averaging the actual signal spectrum with the already averaged flat -96 dB spectrum. The resulting final spectral average will be false.

The recommended ways to operate Repli-Q are:

- Start playback first, and then click the Learn Spectrum button, in that order. Then, click the Learn Spectrum button again to stop learning.
- Alternatively, once the Learn Spectrum button is clicked, start playback immediately, and when

playback is stopped (or ends on its own), click the Learn Spectrum button again immediately to stop the learning process – or digital silence may be averaged into the learned profile.

The second method is best for learning the spectrum of a very short sample – looping short samples while learning is also a useful technique.

When using BIAS Peak as a bost application, a quick and easy way to capture an accurate spectral profile is:

1. Select the portion of the audio file you wish to learn.



2. Click "Learn Spectrum" button in Repli-Q.

3. Choose "Bounce..." from Peak's Plug-Ins menu, (and then click "No" in the bypass/disable dialog).

4. The spectrum has been learned.

5. Turn off Learn mode and undo the bounce – you may now continue working, with the learned spectrum.

Save Spectrum Button

This control saves the current spectrum as a Repli-Q Spectral Profile, and automatically places it in the Reference pop-up menu as the current/active profile.

Reference Pop-Up Menu

This pop-up menu contains all Repli-Q Spectral Profiles created during the current session (i.e., since the plug-in was instantiated) – and also allows loading Spectral Profiles saved in earlier sessions.

Save EQ Button

This button saves the current EQ curve as a Repli-Q EQ Profile, and automatically loads it in the EQ pop-up menu as the current/active profile.

EQ Pop-Up Menu

This pop-up menu contains all Repli-Q EQ Profiles created during the current session (i.e., since the plug-in was instantiated) - and also allows loading EQ Profiles saved in earlier sessions.

Matching Slider/Numerical Field

This control allows variable spectral matching between an input source's profile and a reference file's profile.

By default, the Matching slider is set to 50%.



Sliders can be reset to their default values by Alt/Option-clicking on them.

When numerical fields are in an editable state (i.e., after clicking in them with the mouse), their values may be changed by typing in new values, by using the Up/Down Arrow keys on your keyboard, or by using the scroll wheel on a compatible mouse.

Smoothing Slider/Numerical Field

This control allows variable EQ smoothing, and helps to prevent resonance artifacts.

By default, the Smoothing slider is set to 0%.

Gain Slider/Numerical Field

This control increases or decreases the output level.

By default, the Gain slider is set to 0dB, and has a range of -20 to 20 dB.

Gain Compensation Checkbox

When active, this control compensates for gain differences introduced by an EQ setting, and makes the output level identical to the input level. The presence of a check in this box means that Gain Compensation is active.

Soft Clip Checkbox

When active, this control acts as a soft limiter to ensure that the output level does not clip due to gain differences introduced by an EQ setting. The presence of a check in this box means that Soft Clip is active.

Spectral Profile/EQ View Radio Buttons

These radio buttons allow choosing between viewing of spectral profiles and EQ curves.

Graph Display

The Graph Display operates in two modes, Spectral Profile Mode and EQ Profile Mode. The mode depends on which of the Spectrum/EQ View radio buttons is selected. Various elements of the Graph Display Area change depending on the mode being used.

Spectrum Profile View Mode Display

In Spectrum View Mode, spectral data is shown in the graph. The X-axis is labeled in Hz (Frequencies), and the Y-axis is labeled in dB (decibels). Three different colored lines are plotted in the Spectrum graph, each represents a unique spectral profile.

5

Yellow – Represents the spectral profile of the Reference file, which has the desirable characteristics you are attempting to match, and eventually apply to another file (i.e., which is referred to here as the "Input" file).

Green – Represents the spectral profile of the Input file, which is the file onto which you are attempting to apply the Reference's desirable spectral characteristics.

Blue – Represents the average spectrum of the Output profile (i.e., the actual degree of matching between the Input and the Reference). The



The Graph Display can be toggled between Spectrum and EQ views using the radio buttons just above the display – Spectrum View is shown here



position of the Blue line within the graph is linked to the position of the Matching slider, and indicates the current output spectrum. When the Matching slider is set to 0%, it matches the Input's (Green line) spectral characteristics perfectly. When set to 100%, it matches the Reference source's (Yellow line) spectral characteristics perfectly. When the Matching slider is set somewhere between 0 - 100%, it represents a blend of both the Reference and Input's spectral characteristics.

EQ View Mode Display

In EQ View Mode, the EQ curve is shown in the graph. The X-axis is labeled in Hz (Frequencies), and the Y-axis is labeled in dB (decibels). The frequency data shown in the graph depends on the position of the Matching slider. When the matching slider is set to 0%, the EQ curve is flat at 0 dB and the Output spectrum is identical to the Input spectrum. When it is set to 100%, the EQ provides full matching and the Output spectrum is identical to the Reference spectrum. Note that the Output and Reference spectrums will not be identical when smoothing and edits have been performed on the EQ. When the matching slider is set to a value between 0 - 100%, the EQ is scaled by the same value and the Output spectrum is a blend of the frequency content of both the Input and Reference spectrums.

Editing EQ Filtering Properties

In EQ View Mode, clicking on the graph display allows editing of the EQ curve. Each single edit adds a "peak" or a "hole" in the EQ curve. The user can control the frequency, gain and bandwidth of the edit.

- Click in EQ Graph Displays the current edit parameters: frequency, gain and Q factor (bandwidth).
- Click & Drag in EQ Graph Dragging vertically controls the gain of the edited peak (dragging down produces a negative gain and therefore a hole rather than a peak). Dragging horizontally controls the frequency position of the edited peak. The Q factor (bandwidth of the edited peak) remains fixed.
- Shift-Click & Drag in EQ Graph Dragging horizontally now controls the Q factor (bandwidth) of the edited peak, while dragging vertically modifies the gain. The frequency remains fixed.

Zoom Button

Zooming in or out is accomplished using the Zoom Button, located at the point where the X and Y-axes meet. This control allows a very detailed zoomed-in view, or can be zoomed-out for a "big picture" view. Option-clicking on the Zoom Button resets the zoom level.

- Zoom In Horizontally Click and drag to the right
- Zoom Out Horizontally Click and drag to the left
- Zoom In Vertically Click and drag up
- Zoom Out Vertically Click and drag down

Parameter Automation

Repli-Q's various controls may be automated (i.e. applied dynamically) within hosts that support parameter automation. For more information, please consult your host's documentation on how to automate parameters.

Using Repli-Q

This section contains some basic "hands-on" getting started tips, and introduces some typical Repli-Q workflows.

To Match the Spectrum of an Audio File:

1. Open a Reference file and select the portion to be used as a Reference – the "Reference" file being the one with the desirable spectral characteristics that you wish to apply to another file.

Note that different sections of a Reference file may have different spectral characteristics. If your host program features markers, you may wish to use these to denote specific portions of the Reference file.

- 2. Instantiate Repli-Q, and start playback of the reference file.
- Click the Learn Spectrum button the spectral profile is plotted as Repli-Q learns it.

- 4. Click the Learn Spectrum button again to end the spectral "learning" process.
- Click the Save Spectrum button to save the learned spectral profile – it will automatically be loaded into the Reference pop-up menu as the current/active spectral reference profile. This profile is plotted as a yellow line in the Graph display.
- 6. Open another audio file this will be the material onto which we apply the Reference audio file's desirable spectral characteristics.
- 7. Click the Learn Spectrum button and start playback, the Input file's spectral profile is plotted as a green line in the Graph display.
- 8. Click the Learn Spectrum button again to end the learning process. As soon as the second learning process is stopped, the Output spectrum's profile is immediately calculated (appears as a blue line in the graph display). At this point, the Graph display will show three differently colored plotted lines yellow is the Reference profile whose spectral characteristics will be applied to the Input file when the plugin is active, green is the Input profile, and blue is the output profile. The EQ curve corresponding to the current Input and Reference profiles may be viewed by switching to the EQ Graph Display (by using the EQ View radio button above the Graph Display).
- 9. Use the Matching slider to adjust the degree of matching between the Reference and Input profiles when the Matching slider is at 0%, the Output profile (blue line) coincides with the Input profile (green line), which means that the Input will not be modified. When the slider is at 100%, the Output profile is identical to the Reference profile (yellow line), applying maximum equalization matching to the Input.
- 10. Use the Smoothing slider to smooth the Output profile (blue line) if it presents too many sharp peaks and dips, which can cause undesirable resonance, or inconsistent performance over different sets of reference monitors.

For additional control, the EQ curve's properties may be manually edited – described in additional steps below.

- In the EQ Graph Display, click and drag to modify EQ gain properties (and/or shift-click and drag to modify the EQ's bandwidth filtering properties).
- 12. Repli-Q is now ready for processing play back the input and the new EQ settings will be applied.

At any point after learning the Input and matching/smoothing/editing the EQ, that EQ may be saved with the Save EQ button. This is useful when the same EQ is needed in the future. Saving the EQ is not required for processing the current Input, however, if not saved, the current EQ will be lost when learning a different spectral profile, or de-instantiating the plug-in.

To Process Using a Saved EQ Profile:

- 1. Instantiate Repli-Q and load a saved EQ profile using the EQ pop-up menu.
- Match, smooth and edit the EQ as desired matching will simply scale the EQ between a flat 0dB curve and the loaded EQ profile.
- Open an input file and play it back the loaded EQ profile is applied.

To Use Repli-Q as an Editable Linear Phase EQ:

- 1. Instantiate Repli-Q and select the EQ Graph Display a flat 0dB curve is displayed.
- 2. Play back audio material that needs EQ changes, and perform as many manual edits to the EQ curve as desired until the desired effect is achieved.
- Bounce/apply the plug-in the custom EQ settings are applied.

To Automatically Compensate for Gain Changes:

• Check the Gain Comp checkbox - gain changes

introduced by EQ settings are compensated for automatically, keeping the output level approximately equal to the input level. This is useful in evaluating actual differences in EQ characteristics independently of differences in loudness – as increases in loudness sometimes translate to "better sounding".

To Prevent Clipping the Output Signal:

• Check the Soft Clip checkbox – potential clipping caused by EQ settings that boost output amplitude is now avoided.

To Manually Control Output Level:

- 1. Uncheck the Gain Comp checkbox.
- 2. Use the Gain slider to adjust the output to the desired level.

Conclusion

Now that you are familiar with the various Repli-Q controls, and its basic workflow, please continue along to the next chapter, where you will learn about the Reveal plug-in.

Chapter 6: Reveal



Chapter 6: Reveal

Introduction

Reveal offers an integrated solution to spectral, power, and phase analysis. Its components can be broken down into three categories:

Frequency Analysis Tools

- Oscilloscope
- Spectral Analyzer (Graph & Meter Displays)
- Spectrogram

Level Analysis Tools

- Peak and RMS Level History Display
- Peak and RMS Level Meters

Phase & Pan Analysis Tools

- Lissajous Phase Scope
- Pan Power History Display
- Phase Correlation Display

This combination of tools provides audio engineers with all the elements needed to analyze recordings (or live input), and is indispensable during the audio production process. Audio engineers typically use these types of analysis tools in the following scenarios:

• In the studio, they are used to determine problem

areas in a mix, or to observe how processing or mixing is affecting the audio content.

• The tools may also be used for analyzing real-time live input, in order to analyze room acoustics, and potential phase issues due to speaker placement and speaker tuning.

It takes a combination of audio analysis tools to get a comprehensive understanding of audio content, and Reveal provides them all, in a single easy-to-use plug-in.

Reveal User Interface & Controls

Reveal is designed to be intuitive and easy-to-use. This section discusses the graphical user interface, and will familiarize you with each of the various controls.

Global Controls

Reveal includes many controls designed for specific tasks, and also features several "global" controls, that affect the state of the entire plug-in. Below is a brief description of these "global" tools, and how they operate.

57

Reveal Graphical User Interface



*Appearance of preset/bypass controls will vary depending on the platform, host, and plug-in format used

Resolution Pop-Up Menu

This control sets the screen update rate used in Reveal's various displays. The resolution may be adjusted to suit the CPU of the computer on which Reveal is being used. Higher resolution settings are intended for faster CPUs, and lower settings are intended for slower CPUs – you may want to experiment with this setting to get the best results for your particular computer system.

Global Freeze Button

Clicking the Global Freeze button temporarily stops movement in all displays, allowing for closer examination of their data. Clicking this button again resumes activity in all displays.

Input Source Pop-Up Menu

This menu allows you to choose whether the input source (i.e., the signal being fed into Reveal for analysis) is a single (left or right) channel of a stereo file, or both channels simultaneously.



Zoom Buttons

Technically, these are not "global" controls, but they appear in several of Reveal's tools, so they will be covered here.



The Zoom Button appears in several of Reveal's tools, and allows zooming along the X & Y-axes.

Zooming and scrolling is possible within tools that feature this control, which allows a very detailed zoomed-in view, or can be zoomed-out for a "big picture" view. Zooming is accomplished using the Zoom Button, located at the point where the X and Y-axes meet. Option-clicking on the Zoom Button resets the zoom level.

- Zoom In Horizontally Click and drag to the right
- Zoom Out Horizontally Click and drag to the left
- Zoom In Vertically Click and drag up
- Zoom Out Vertically Click and drag down

A/B/C/D Compare Buttons

The A/B comparison feature is common in audio programs, and gives the ability to preview different settings before committing to them. Essentially, various tool settings are made and stored as snapshot "A" another group of settings can then be assigned as snapshot "B" – and then clicking the A or B buttons in the interface instantly toggles between the two. As Reveal offers many fine-tuning options, it features not just two comparison modes - but instead includes A/B/C/D comparisons, for four unique snapshots to compare between, before committing to any file processing. These snapshots are similar to saving and loading presets, but are instantaneous, and do not require selecting presets from a menu to load them. To permanently save one of these snapshots, simply select the desired snapshot, and save it as a custom User Preset.

Factory Preset Pop-Up Menu

This menu displays the current factory preset, if one has been selected. Clicking this menu shows a list of all available factory presets. A factory preset may be selected by choosing it from this pop-up menu.

Factory Preset Selection Arrows

These arrows are used to navigate through factory presets. Clicking the down arrow moves to the next available factory preset, and clicking the up arrow moves to the previous factory preset.

User Preset Save Button

Saves a user preset, which includes the state of all Reveal parameters.

User Preset Load Button

Recalls a user preset, which includes the state of all Reveal parameters.



User presets do not appear in the Factory Preset pop-up menu, be aware of the directory where you are saving them into, so they may be recalled easily later.

Bypass Button

The Bypass button is used to temporarily disable the plug-in. When the plug-in is bypassed, audio playback is not analyzed by the plug-in.

Tool-Specific Controls & Displays

While all of Reveal's analysis tools may be viewed simultaneously in the View All Displays Tab, you may also view each tool independently. Viewing a single tool offers a much larger display area, which can translate into more accurate data interpretation. Having this option allows you to view many types of data simultaneously, or a single type of data in great detail.

Navigating between the various tools is possible by clicking the desired tool's tab in the lower part of the Reveal interface will do the same. In this section, we'll take a look at each of Reveal's tools.

Oscilloscope

The Oscilloscope displays a trace-style scope waveform. Left channel data is displayed in green, and right channel data is displayed in red. The graph display's X-axis measures amplitude, and is labeled in dB (decibels), while the Y-axis measures time, and is labeled in samples.



Reveal's Oscilloscope Display



Peak and RMS Power History Tool

This tool analyzes an audio file's Peak and RMS ("Root Mean Square" or average) level history. By plotting the Peak and RMS levels in a graph it's possible to monitor the volume of an audio signal over a period of time. The data plotted in the graph display area includes: The graph display's Y-axis measures amplitude, and is labeled in dB (decibels), while the X-axis measures time, and is labeled in minutes and seconds. The precision of the intervals along each axis varies, and depends on the zoom level being used.

Spectrogram Tool

- Left Channel Peak Level Orange
- Right Channel Peak Level Red
- Left Channel RMS Level Yellow
- Right Channel RMS Level Pink

The Spectrogram analyzes an audio file's spectral content, and displays it in a time-frequency graph (in which the X-axis represents time and the Y-axis represents frequency). A color gradient is used to represent the varying amount of power present in



Chapter 6: Reveal

different parts of the frequency spectrum. Red represents the highest amplitude for a given frequency, while green represents the lowest amplitude.

Analysis Window Slider/Numerical Field

The Analysis Window control allows varying frequency resolution versus time (attack/decay) resolution. The higher the Analysis Window setting, the more frequency resolution and the less time resolution is displayed, and vice-versa.

A higher Analysis Window setting means that partials/ harmonics (i.e., sine waves) with very close frequencies will be seen as separate peaks in the spectrum (each peak is a lobe about 7 spectral points wide). If the Analysis Window setting is steadily decreased, at some point the peaks will merge into one large lobe, and it will be impossible to distinguish whether separate partials exist.

A lower Analysis Window setting means that two attacks which occur very close to each other in time will be seen as two separate events (i.e., the spectrum will jump up, decay, and jump up again). If the Analysis Window setting is steadily increased, at some point the two attacks will be within one Analysis Window, and the spectrum will only go up once.

Therefore, if partials/harmonics are important for the analysis you are performing, the Analysis Window setting should be increased until separate spectral peaks are visible. Too big of an Analysis Window averages too much of the signal, so fast changes (attacks or fast pitch variations) merge into one blurry spectrum.

For monophonic signals, the Analysis Window setting should be around 4 periods of signal (on average, since the pitch is changing) for adequate frequency resolution (i.e. so that the partials are seen as separate peaks in the spectrum).

By default, the Analysis Window slider is set to 50ms, and has a range of 10 - 100ms.

The graph display's Y-axis measures frequency, and is labeled in Hz (Hertz), while the X-axis measures time, and is labeled in minutes and seconds. The precision of

the intervals along each axis varies, and depends on the zoom level being used.



Sliders can be reset to their default values by Alt/Option-clicking on them.

When numerical fields are in an editable state (i.e., after clicking in them with the mouse), their values may be changed by typing in new values, by using the Up/Down Arrow keys on your keyboard, or by using the scroll wheel on a compatible mouse.

Pan Power/Phase Correlation Tool

This tool analyzes phase correlation and power balance between a stereo signal's left and right channels. Phase correlation data is shown in the Phase Correlation Meter. Maximum correlation (i.e., a strong "In Phase" reading) means the left and right signals are identical. Zero correlation (i.e. meter reading is centered) means the signals are completely unrelated (i.e., panned hard left or right). Pan power data is shown in the Pan Power Graph. If both channels contain equal power, the meter reading is centered. If one channel contains more power than the other, the meter reading appears stronger for that channel.

The Phase Correlation meter displays correlation as a percentage, either in phase or out of phase.

The Pan Power graph display's X-axis measures pan power in decibels (left/right), while the Y-axis measures time, and is labeled in minutes and seconds. The precision of the intervals along the Y-axis varies, and depends on the zoom level being used.

Spectral Analysis Tool

The Spectral Analysis tool analyzes frequency content contained in an audio file, and displays it in the form of a



Spectral Meter or Spectral Graph, with amplitude on the Yaxis, and frequency on the X-axis. Left channel data is displayed in green, and right channel data is displayed in red.

Display Pop-Up Menu

This control allows configuring the Spectral Meter or Graph to display Instantaneous, Peak, or Average levels.

Response Time Slider/Numerical Field

The Response Time control is used to set the duration over which the Average Spectrum is calculated. A low

setting causes it to react relatively quickly, and a high setting causes a slower reaction.

By default, this slider is set to 50ms, and has a range of 10 - 100ms.

Analysis Window Slider/Numerical Field

The Analysis Window control sets the amount of data being analyzed, and allows varying frequency resolution versus time (attack/decay) resolution. The higher the Analysis Window setting, the more frequency resolution and the less time resolution is displayed, and vice-versa.



Chapter 6: Reveal

6



Reveal's Spectral Analysis Tool – showing the Spectral Graph Display

A larger Analysis Window will better separate partials/harmonics, but will also average spectral variations within the duration of the window.

A shorter Analysis Window will better separate spectral variations over time (i.e., will more clearly show an attack of the piano as a sudden rise of the middle portion of the spectrum), but will not separate partials as well, and eventually, with a short enough setting, discrete spectral peaks for each partial will not be visible.

A good rule of thumb is to make this window three to four times the average pitch period in the signal. Some common examples include:

- Female vocals (around 300 Hz pitch), the window should be around 4/300, or 0.01333 seconds, or 13.33 milliseconds long.
- Bass (around 50 Hz), the window should be around 4/50 or 0.08, or 80 milliseconds long.
- Complete Mix Use the lowest-pitch instrument (usually the bass) as the guideline for setting the Analysis Window size.

The graph display's X-axis measures frequency, and is labeled in Hz (Hertz), while the Y-axis measures amplitude, and is labeled in dB (decibels).

Lissajous Phase Scope Tool

The Phase Scope tool displays the amplitude and phase of two signals (i.e., the left and right channels of a stereo signal) as a Lissajous pattern, in which the left channel's frequency data is plotted along the graph's Y-axis, and the right channel's is plotted along the X-axis. Lissajous patterns are useful for determining phase and amplitude relationships between the left and right signals of a stereo pair.

Peak/RMS Level Meters Tool

This tool contains both Peak and RMS ("Root Mean Square" or average) Level metering. Other features include Clip Indicators, Clip Statistics, and variable Peak Hold, RMS Smoothing, and Meter Range. Each of these features is described below.

Clip Indicators

These indicators are located just above the Peak Level Meters. They light up in red when the signal exceeds 0dBFS. The Clip Indicators can be reset by clicking on them.

Peak Level Indicators

These indicators appear as horizontal yellow lines, at the top of the Peak Level Meter display. Various options for these indicators may be set using the Peak Hold pop-up menu.



Overall Peak Level Indicator

This indicator displays the overall Peak Level, with an infinite hold time (similar to setting the Peak Hold time for the Peak Level indicators to Infinity). This indicator can be reset by clicking the Reset All button.

Sample Count in Last Clip Display/Button

This control is both a numeric display, and a button - it shows how many samples were clipped in the last clipping incident. Clicking this button resets the clipped sample count to zero.

Total Samples Clipped Display/Button

This control is both a numeric display, and a button - it shows the total number of samples in all clipping incidents. Clicking this button resets the clipped sample count to zero.

Total Clips Display/Button

This control is both a numeric display, and a button - it shows the total number of clips. Clicking this button resets the clip count to zero.



6

Peak Hold Pop-Up Menu

This control sets the duration that the Peak Level indicators stay at the Peak Level.

By default, this control is set to 3 Seconds, and options include:

- Off
- 1 Second
- 3 Seconds
- 5 Seconds
- Infinity

RMS Smoothing Pop-Up Menu

This control sets the duration over which RMS level values are calculated. By default, this control is set to 3 Seconds, and options include settings between 1 - 10 Seconds.

Meter Range Pop-Up Menu

This control sets the range of both the Peak and RMS meters. By default, this control is set to -96dB, and options include:

- -48dB
- -96dB
- -144dB

Parameter Automation

Reveal's various controls may be automated (i.e. applied dynamically) within hosts that support parameter automation. For more information, please consult your host's documentation on how to automate parameters.

Conclusion

Now that you are familiar with the various analysis tools in Reveal, continue along to the next chapter, which covers SuperFreq.

Chapter 7: SuperFreq



Chapter 7: SuperFreq

Introduction

SuperFreq is a high-end suite of paragraphic equalizer plug-ins, which includes 4-, 6-, 8-, and 10-band editions, so the right amount of EQ control is always available.

SuperFreq includes everything for professional-caliber equalization, including –24dB to +24dB Gain Values; 0.1 to 30 "Q" (bandwidth) values; sweepable frequencies from 20 Hz to 20 kHz, individual band bypass buttons; 24 dB stereo input/output meters; and filters including Peak, Notch, High Shelf, Low Shelf, High Cut, and Low Cut.

SuperFreq is now at version 2.0, and includes the following new features:

- Updated User Interface
- Numerical fields allow parameter values to be typed in, nudged up or down with arrow keys, or modified using a compatible scroll wheel mouse
- A/B/C/D Compare buttons
- Wider Gain value range (+/- 24dB)
- High precision, high contrast meters
- RIAA Compensation preset

SuperFreq User Interface & Controls

The SuperFreq interface is similar for the four SuperFreq plug-ins (SuperFreq-4, SuperFreq-6, SuperFreq-8, and SuperFreq-10). The number of vertical sets of knobs in

the middle of the interface corresponds to the number of bands of EQ — for example, SuperFreq-10 has 10 bands of EQ, so its interface has 10 vertical sets of knobs in the middle of the interface.

Global Controls

SuperFreq features a variety of controls designed for specific tasks, and also features a number of "global" controls, that affect the state of the entire plug-in. Below is a brief description of these "global" tools, and how they operate.

A/B/C/D Compare Buttons

The A/B comparison feature is common in audio programs, and gives the ability to preview different settings before committing to them. Essentially, various tool settings are made and stored as snapshot "A" another group of settings can then be assigned as snapshot "B" – and then clicking the A or B buttons in the interface instantly toggles between the two. As SuperFreq offers many fine-tuning options, it features not just two comparison modes - but instead includes A/B/C/D comparisons, for four unique snapshots to compare between, before committing to any file processing. These snapshots are similar to saving and loading presets, but are instantaneous, and do not require selecting presets from a menu to load them. To permanently save one of these snapshots, simply select the desired snapshot, and save it as a custom User Preset.

SuperFreq 2.0 Graphical User Interface



*Appearance of preset/bypass controls will vary depending on the platform, host, and plug-in format used

Factory Preset Pop-Up Menu

This menu displays the current factory preset, if one has been selected. Clicking this menu shows a list of all available factory presets. A factory preset may be selected by choosing it from this pop-up menu.

Factory Preset Selection Arrows

These arrows are used to navigate through factory presets. Clicking the down arrow moves to the next available factory preset, and clicking the up arrow moves to the previous factory preset.

User Preset Save Button

Saves a user preset, which includes the state of all parameters.

User Preset Load Button

Recalls a user preset, includes the state of all parameters.



Bypass Button

The Bypass button is used to temporarily disable the plug-in. When the plug-in is bypassed, audio playback is not affected.



The Factory, User Preset, and Bypass controls are found in a hostprovided section of the interface. This will vary depending on the host and plug-in format being used.

SuperFreq Factory Presets

SuperFreq provides several factory loaded presets to help get you started: Default, Bass Boost, Loud, SuperLoud, Double Low Cut, DeNoiser, Comb Filter, Phony, and RIAA Phono Compensation.

Default

Sets all parametric frequency bands to 0 dB.

Bass Boost

Provides a Low-Shelf boost of 6 dB starting at 130 Hz.

Loud

Provides a moderate boost (4 dB) to the high-end (High-Shelf at 7 kHz) and low-end (Low-Shelf at 200 Hz), while slightly attenuating (–2 dB) the mid-range (Peaking at 2 kHz).

SuperLoud

Provide a prominent boost (7 dB) to the high-end (High-Shelf at 7 kHz) and low-end (Low-Shelf at 200 Hz), while slightly attenuating (–2 dB) the mid-range (Peaking at 2 kHz).

Double Low Cut

Rolls off the low-end using two Low-Cut filters at 60 Hz and 40 Hz respectively.

DeNoiser

Rolls off the high-end using a High-Cut filter at 10 kHz and a few notch filters (Peaking) to eliminate frequencies at 60 Hz, 120 Hz, and 180 Hz. This is a useful setting to help reduce room noise.

Comb Filter

Provides as many notch filters (Peaking) as there are

Parametric Filter Bands available in the specific version of SuperFreq (i.e., 4 in SuperFreq-4 and 10 in SuperFreq-10). A comb filter can produce some very interesting special effects.

Phony

Eliminates the high-end and low-end by using a gentle Low-Cut filter at 500 Hz and a gentle High-Cut filter at 3 kHz. The result is that only the mid-range frequencies pass through, not unlike the bandwidth of a telephone signal.

RIAA Phono Compensation

This preset compensates for RIAA equalization that is inherent in the record manufacturing process (in which low frequencies are cut and high frequencies boosted). This preset boosts the low frequencies, and cuts the high frequencies. When vinyl records are digitally recorded without a suitable phono preamplifier, their RIAA EQ curve must be compensated for, in order for the material to sound "normal".

Output Level Controls

The Output Gain control and Output meters are located on the far right of the plug-in window. This gain control allows you to attenuate or boost the gain of the signal coming out of the plug-in. However, if you are significantly altering the frequency content of a signal, there can be a significant difference in the output gain from the input gain. Use the output gain slider to reduce the gain to prevent clipping. If the output is clipping, the meters will light the top light of the meter, known as the clip light. This light lets you know that something has clipped even after it has happened. To turn off the clip lights, click on one of them.

Equalization Controls

A set of master compression controls appears just above the Frequency Graph Display area. These controls affect individual equalization bands. Each control is described in further detail below.

Parametric Filter Bands

The vertical sections that consist of a pop-up menu and three vertically aligned knobs are the individual filter bands. Each one of these bands is independent and can alter arbitrary parts of the frequency spectrum.



Parametric Filter band with controls for gain, frequency, and Q

Filter Type Pop-Up Menu

SuperFreq features five types of parametric filters: Peak, Low Shelf, High Shelf, High Pass, and Low Pass. Also listed in the Filter Type pop-up menu is an item called Thru. Choosing Thru has the same effect as clicking the filter band's Bypass checkbox. This is helpful when evaluating the sound difference of a specific band.



Multiple filter types are available for each band
A Peak filter is the classic parametric filter. It allows you to boost or cut a specific area of frequencies with an arbitrary bandwidth.

The Low Shelf and High Shelf filters boost or cut all the frequencies below or above the selected frequency. These are useful for boosting the bass or bringing up the detail in the treble regions.

The High Pass and Low Pass filters cut out frequencies above or below the specified frequency. These are mostly used to eliminate high-frequency noise or lowfrequency rumble.

Boost/Cut Knob

The knob right below the Filter Type pop-up menu is the Boost/Cut knob. This knob attenuates or boosts the signal gain in dB for the specified frequency band. This knob has no effect for the Low Cut and High Cut filters.

Frequency Knob

The knob below the Boost/Cut knob controls the frequency at which the filter is applied. For a Peaking filter, it determines the frequency of the center of the peak or notch. For a Low Shelf or High shelf filter it determines the frequency of the transition. For High Pass and Low Pass filters it determines the frequency at which the filters start to cut out the signal.

Q Knob

The bottom knob controls the Q, or bandwidth, of the filter. For a Peak or Pass filter this knob controls the width of the peak or notch. For the Shelf and Pass filters it controls the transition region of the filter — a standard value for high cut and low cut filters is 0.7. Higher Q values create a steeper transition between the boosted or cut frequencies and the unaffected frequencies.

Bypass Checkbox

The checkbox labeled Bypass at the bottom of each filter

band toggles the band on or off. This is helpful when evaluating the sound difference of a specific band.



Knobs & sliders can be reset to their default values by Alt/Option-clicking on them.

When numerical fields are in an editable state (i.e., after clicking in them with the mouse), their values may be changed by typing in new values, by using the Up/Down Arrow keys on your keyboard, or by using the scroll wheel on a compatible mouse.

Frequency Graph Display

The Frequency Graph display provides information about all frequency bands and allows you to change each band's gain and frequency values. The graph represents the entire range of human hearing, from 20Hz on the left, to 20kHz on the right.



SuperFreq's Graph Display

The horizontal line through the middle of the display represents the dry signal passing straight through the plug-in (without altering any frequencies in the signal).

The light blue region on the graph represents changes to the signal's frequency content. If the blue region is above the horizontal line you are boosting those frequencies, if it is below you are cutting them.

The vertical divisions on the graph are 8 dB, so the graph can display up to +24 dB and down to -24 dB. You can also control some of the band parameters from here. All of the active bands display a little white ball with their corresponding band number on them. Click the

ball for the desired band and drag it left or right to change the band's frequency parameter and drag it up or down to change its boost/cut parameter.

Parameter Automation

SuperFreq's various controls may be automated (i.e. applied dynamically) within hosts that support parameter automation. For more information, please consult your host's documentation on how to automate parameters.

Using SuperFreq

This section covers the basics of using SuperFreq. It will teach you how to open, make parameter adjustments to, apply, and close SuperFreq.

To Boost a Frequency Band:

- 1. Open SuperFreq.
- 2. Select a frequency band to boost.
- 3. Click the band's ball in the frequency graph, and move it upwards (or turn the band's gain knob to the right/clockwise).

To Cut a Frequency Band:

- 1. Open SuperFreq.
- 2. Select a frequency band to cut.
- 3. Click the band's ball in the frequency graph, and move it downwards (or turn the band's gain knob to the left/counter-clockwise).

To Adjust the Center Frequency of a Filter Band:

- 1. Open SuperFreq.
- 2. Select a band in which to adjust the center frequency.
- 3. Click the band's ball in the frequency graph, and move it to the left or right – or use the frequency knob (Hz) to make this adjustment. Moving the filter band's ball in the graph to the left (or turning the Hz knob to the left) moves the center frequency lower in the frequency spectrum. Moving the filter band's ball to the right (or turning the Hz knob to the right) moves the center frequency higher in the spectrum.

To Adjust Bandwidth:

- 1. Open SuperFreq.
- 2. Select a band to adjust its bandwidth (how wide a range of frequencies it affects).
- 3. Turn the Q knob for that filter band to the left/counterclockwise, to create a wide bandwidth – or turn it to the right/clockwise to create a narrow bandwidth. Smaller values represent wide bandwidth, and larger values represent narrow bandwidth.

To Change a Band's Filter Type:

- 1. Open SuperFreq.
- 2. Select a band to change its filter type.
- 3. Click on the Filter Type pop-up menu, and choose the desired filter type. Descriptions of filter types appear in the section "SuperFreq User Interface & Controls".

To Boost/Cut Output Gain:

- 1. Open SuperFreq.
- 2. Make adjustments to desired filter bands.
- If you lose overall gain with the adjustments made to individual frequency bands, raise the output gain slider to make up for lost gain – and if you raised the overall gain with the individual band

adjustments, lower the output gain slider to reduce overall gain.

To Bypass a Filter Band:

- 1. Open SuperFreq.
- 2. Make desired settings.
- Click the Bypass checkbox for the band you wish to temporarily bypass – while the Bypass checkbox is checked, that filter does not affect audio passing through SuperFreq.

Conclusion

Now that you have learned a bit about SuperFreq, please continue along to the next chapter, where you will learn about the Sqweez-3 & -5 multiband compressor plug-ins.

Chapter 8: Sqweez-3 & -5



Chapter 8: Sqweez-3 & -5

Introduction

Sqweez-3 & -5 represent the very latest in professional multi-band compression, limiting, and upward expansion technology. Its linear phase filters give precise control over compression settings for three frequency bands in the case of Sqweez-3, and five bands with Sqweez-5. The advanced visual display provides crucial analysis feedback, and allows parameter adjustment with the highest degree of control.

Sqweez-3 & -5 allow independent compression settings to be applied for low, mid, and high frequencies, and each band can be configured to affect the exact frequency range desired.

This chapter will introduce the graphical user interfaces, how the controls work, and gives some basic hands-on getting started tips for working with Sqweez-3 & -5.

Sqweez-3 & -5 User Interface & Controls

Sqweez-3 & -5 are designed to be intuitive and easy to use. This section discusses the graphical user interface, and will familiarize you with each of the various controls.

Global Controls

Sqweez-3 & -5 feature a variety of controls designed for specific tasks, and also features a number of "global"

controls, that affect the state of the entire plug-in. Below is a brief description of these "global" tools, and how they operate.

A/B/C/D Compare Buttons

The A/B comparison feature is common in audio programs, and gives the ability to preview different settings before committing to them. Essentially, various tool settings are made and stored as snapshot "A" another group of settings can then be assigned as snapshot "B" - and then clicking the A or B buttons in the interface instantly toggles between the two. As Sqweez-3 & -5 offer many fine-tuning options, it features not just two comparison modes - but instead includes A/B/C/D comparisons, for four unique snapshots to compare between, before committing to any file processing. These snapshots are similar to saving and loading presets, but are instantaneous, and do not require selecting presets from a menu to load them. To permanently save one of these snapshots, simply select the desired snapshot, and save it as a custom User Preset.

Factory Preset Pop-Up Menu

This menu displays the current factory preset, if one has been selected. Clicking this menu shows a list of all available factory presets. A factory preset may be selected by choosing it from this pop-up menu.

Factory Preset Selection Arrows

These arrows are used to navigate through factory

Sqweez-3 Graphical User Interface



*Appearance of preset/bypass controls will vary depending on the platform, host, and plug-in format used

presets. Clicking the down arrow moves to the next available factory preset, and clicking the up arrow moves to the previous factory preset.

User Preset Save Button

Saves a user preset, which includes the state of all parameters.

User Preset Load Button

Recalls a user preset, includes the state of all parameters.



User presets do not appear in the Factory Preset pop-up menu, so be sure you are aware of the directory where you are saving them into, so they may be recalled easily later.

Bypass Button

The Bypass button is used to temporarily disable the plug-in. When the plug-in is bypassed, audio playback is not affected.

Master Compression Controls

A set of master compression controls appears just above the EQ/Threshold Graph Display area. The controls affect all compression bands. Each control is described in further detail below.

Gain Compensation Button

When clicked, this control toggles automatic gain compensation on or off. When Gain Compensation is active, the output signal level is approximately equal to the level of the original input signal.

Spectrograph View Pop-Up Menu

This control allows choosing whether left channel, right channel, or stereo channel information is shown when the Spectrograph is active. The spectrograph display appears only in the Threshold View (or in the Threshold portion of the "Both" View).

Spectrograph Mode Pop-Up Menu

This control allows the Spectrograph display to be turned on or off. Two display options are available when the Spectrograph is active – Instantaneous and History. Instantaneous mode displays the current spectral qualities of the audio signal. History mode displays an RMS Average of the file's spectrum over the past few seconds.

Spectrograph Pre/Post Pop-Up Menu

This control allows the display of spectral data in a preprocessing or post-processing state. It is useful for visually comparing the "before" and "after" states of compression settings before the plug-in is applied.

Soft Clip Button

When active, this control acts as a soft limiter to ensure that the output level does not clip due to gain differences introduced by a compression setting. The presence of a check in this box means that Soft Clip is active.

Digital Ceiling Slider/Numerical Field

This control sets the maximum allowable level in decibels (used by the Soft Clip control's algorithm).

By default the Digital Ceiling slider is set to 0dB, and has a range of -20 to 0dB.

8



Sliders/pop-up sliders can be reset to their default values by Alt/Option-clicking on them.

When numerical fields are in an editable state (i.e., after clicking in them with the mouse), their values may be changed by typing in new values, by using the Up/Down Arrow keys on your keyboard, or by using the scroll wheel on a compatible mouse.

Input/Output Meters

These meters show levels for both the original signal being input, and the output level, after being affected by the current compression parameters.

Master Gain Slider

This control affects output level globally.

By default, the Gain Slider is set to a value of 0dB, and has a range of -12 to 12 dB.

Graph Display

This display shows a variety of compression data for each band in graphical format. The Graph display may be viewed in three modes – EQ, Threshold, or Both (which displays EQ & Threshold data simultaneously). The graph's X-axis represents frequencies and the Y-axis represents amplitude. In Threshold or Both mode, a spectrograph may be optionally displayed. The Spectrograph can be especially useful for visualizing the frequency range of independent sources in the mix, and setting crossover frequencies accordingly (see description of Spectrograph, earlier in this chapter).

The Graph display presents many useful pieces of

information in a way that is easy to interpret with just a quick glance. The display's elements are described below.

Display Mode Radio Buttons

This set of radio buttons allows choosing between the three available display modes – EQ, Threshold, or Both.

Filter Crossover Indicator/Numerical Entry Field

The Filter crossovers appear in the display as vertical lines, and are used to adjust the width of each compression band (and edit the crossover frequency between it and its neighboring bands). Each crossover indicator features a control tab at its top and bottom, and a numerical entry field so that specific custom values may be quickly entered. Numerical entry fields are staggered so as to fit the interface smoothly – the field for the crossover between bands 1 and 2 appears at the bottom of the indicator, the field for the crossover between bands 2 and 3 appears at the top of the indicator, and so on.

Threshold Indicator

The Threshold indicators appear only in Threshold or Both view modes, and are used to adjust the compression threshold for each band. Each band's Threshold indicator corresponds (and is linked) to the Threshold slider in the band-specific control area below the graph display area – moving the indicator in the graph also moves the slider, and vice-versa.

Gain Indicator

The Gain level for each band is indicated by a thin, light blue line (see Graphical User Interface overview earlier in chapter) in the EQ graph display area. This indicator appears in a horizontal position by default (with all bands set to 0dB, but its shape and orientation change with adjustments made to the Gain slider and Maximum Compression Indicator). The Gain Indicator appears in the EQ and Both view modes.

Maximum Compression Indicator

This indicator displays the maximum amount of compression currently being applied. It appears as a thin white line (see Graphical User Interface earlier in chapter) in the EQ graph display area. The Maximum Compression Indicator appears in the EQ and Both view modes.

Current Compression Indicator

This indicator displays the compressor's current frequency response at any given instant. It appears as a thin yellow line (see Graphical User Interface overview earlier in chapter) whose position will vary depending on the current compression settings – but will never exceed the amount of compression set with the Maximum Reduction pop-up slider. The Current Compression Indicator appears in the EQ and Both view modes.

Gain Edit Control

The Gain Edit controls appear as small white threedimensional balls within each compression band in the graph display. These controls are used to adjust a compression band's Gain level, and they correspond with (and are linked to) the Gain slider in the bandspecific control area below the graph display area – moving the Gain Edit control in the graph display also moves that band's Gain slider, and vice-versa. The Gain Edit controls appear in EQ or Both View mode.



Each band contains an upper and lower control ball. The upper one controls Gain, and the lower one controls the Maximum Reduction setting (see next control description).

Maximum Reduction Edit Control Balls

The Maximum Reduction Edit Controls appear as small white three-dimensional balls within each compression band in the graph display. These controls are used to adjust a compression band's Maximum Reduction, and they correspond with (and are linked to) the Maximum Reduction pop-up slider in the band-specific control area. Moving the Maximum Reduction Edit control in the graph display also moves that band's Maximum Reduction popup slider, and vice-versa. The Maximum Reduction Edit controls appear in EQ or Both View mode.

Band-Specific Controls

In addition to the global control discussed in the previous section, band-specific controls are also included. These allow fine-tuning of particular compression characteristics that need to be made only within a certain frequency range.

Threshold Slider

This control allows setting a threshold level – audio signals that exceed this threshold are processed with the current compression settings. Signals that fall below the threshold are not affected by the band's compression settings.

By default, Threshold Sliders are set to 0dB, and have a range of 0 to -90dB.

Threshold Input Meter

Each band features an input level meter, which is specific to the frequencies that fall within the range of that band. These meters are useful for setting the Threshold Sliders within each band.

Gain Slider

This control can be used to manually make up gain that may be lost as a result of compression settings within a given band.

By default, the Gain Sliders are set to 0dB, and have a range of -12 to 12dB.

Ratio Pop-Up Slider

This control determines how much a signal is reduced when it exceeds the threshold level for that band. For example, using a 3:1 ratio means that every 3dB over the threshold level results in a signal that is 1dB over the threshold, which is a reduction of 2dB.

By default, the Ratio Pop-Up Sliders are set to a value of 2:1, and have a range of 2:1 to 50:1.

Maximum Reduction Pop-Up Slider

This control allows setting a maximum reduction value for a given band. Some ratio and threshold settings may result in unrealistic reduction values. This slider allows setting a user-defined maximum amount of reduction.

By default, the Maximum Reduction sliders are set to 0dB, and have a range of 0 to 60dB.

Attack Pop-Up Slider

This control determines how quickly compression is applied, when a signal exceeds the threshold for that band.

By default, the Attack Pop-Up Sliders are set to 30ms, and have a range of 0 to 200ms.

Release Pop-Up Slider

This control determines how quickly compression effect is released, once a signal falls below the threshold level for that band.

By default, the Release Pop-Up Sliders are set to 200ms, and have a range of 0 to 200ms.

Knee Pop-Up Slider

This control determines how hard (dramatic) or soft (transparent) the knee of the compression curve is at the threshold point.

By default, the Knee Pop-Up Sliders are set to .50dB, and have a range of .50 to 3.00dB. (This value corresponds to the interval on which the compression curve is smoothed around the threshold.)

Bypass Checkbox

The Bypass checkbox is used to temporarily disable a specific compression band. When a band is bypassed, audio within its frequency range is not affected by compression settings. The presence of a check means that band is bypassed.

Solo Checkbox

The Solo checkbox is used to temporarily disable all other compression bands. When a band is soloed, only audio within its frequency range is affected by compression settings. The presence of a check means that band is soloed.

Group Controls

This set of controls is used to apply common Ratio, Maximum Reduction, Attack, Release, and Knee settings to all bands simultaneously.

Parameter Automation

Sqweez-3 & -5's various controls may be automated (i.e. applied dynamically) within hosts that support parameter automation. For more information, please consult your host's documentation on how to automate parameters.

Using Sqweez-3 & -5

This section contains some basic "hands-on" getting started tips for using Sqweez-3 & -5.

To Apply Compression (Basic Steps):

- 1. Instantiate Sqweez-3 or -5, open an audio file, and start playback.
- 2. In each compression band, use the input meter to help gauge the appropriate Threshold level and set this using the Threshold slider. The amount of compression applied depends on how far the Threshold slider is set below the meter's highest reading.
- 3. Click the Gain Compensation button (or use the Master Gain Slider) to make up for overall output level reduction introduced by the compression settings.
- 4. In each band:
 - Adjust the Ratio, Attack and Release to suit your audio material.
 - Adjust the Gain slider to a level that suits the content in that band.

To Fine-Tune Compression Settings:

1. Click and drag the Filter Crossover Indicators to create compression bands that span frequency ranges best suited for the audio material you are working with.



The Spectrograph Display can help to visualize the audio material's frequency spectrum, and is useful in determining the ideal position for each filter crossover.

- 2. Check the Soft Clip Checkbox to prevent unwanted clipping.
- 3. Use each band's Bypass and Solo controls to audition changes to each band's settings.

- 4. For a more dramatic compression effect, try the following:
 - Decrease the Attack or Release values.
 - Set the Knee to a higher value.
 - Set the Ratio to a higher value.
 - Increase the Maximum Reduction level.

Conclusion

You should now be familiar with the various plug-ins in the Master Perfection Suite, and their basic operation. For more information, or for additional help, please visit the BIAS website, where you can find tutorials, access technical support, and search through a database of frequently asked technical questions.

We hope you enjoy the Master Perfection Suite!